On local connectivity for the Julia set of rational maps: Newton's famous example

By P. Roesch

Abstract

We show that Newton's cubic methods (famous rational maps) have a locally connected Julia set except in some very specific cases. In particular, when those maps are infinitely renormalizable their Julia set is locally connected and contains small copies of nonlocally connected quadratic Julia sets. This also holds when Newton's method is renormalizable and has Cremer points, unlike the polynomial case. After a dynamical description we show the necessity of the Brjuno condition within this family.

1. Introduction

In this article we are interested in the local connectedness of the Julia set of rational maps acting on the Riemann sphere $\widehat{\mathbf{C}}$. This problem is central in holomorphic dynamics, notably in order to obtain topological models and to approach the famous MLC conjecture; it has been studied a lot in the case of polynomials (see [D-H1], [F], [G-Sm], [G-Sw], [K], [L-vS], [Ly], [McM1], [Pe1], [Ra], [Ri], [So1], [So2], [T-Y],...), and mostly in degree two — but hard questions still remain. Important progress was made by Yoccoz who proved that if a quadratic polynomial has only repelling periodic points (in \mathbf{C}) and is not infinitely renormalizable, then its Julia set is locally connected (see [Hu], [M2]). Douady exhibited then striking examples of infinitely renormalizable polynomials having a nonlocally connected Julia set (see [M2] and also [So2]). Several years before, Sullivan had given the first examples of such pathological Julia sets by showing that every polynomial with a Cremer point has a nonlocally connected Julia set (see [M1]). A *Cremer* point is a periodic point in the Julia set whose first return map is tangent to an irrational rotation. As we will see here, rational maps may behave in a completely different way:

THEOREM 1. There exist rational maps which have a locally connected Julia set and Cremer points; cubic Newton maps provide such examples.

The question is whether all cubic Newton maps with a Cremer point have a locally connected Julia set is still open (see Question 8.5 and Conjecture 8.6). On the other hand, THEOREM 2. Every infinitely renormalizable¹ genuine cubic Newton map has a locally connected Julia set.

A genuine Newton map will be a Newton map that is not quasi-conformally conjugated to a polynomial in a neighborhood of its Julia set.

Cubic Newton maps arise as natural examples to look at: besides quadratic, they are the only rational maps with simple critical points which are all fixed except one which is "free" (see Lemma 2.1); moreover they form a family of dynamical systems in which this critical point displays all possible behaviors.

COROLLARY 3. There exist rational maps with locally connected Julia sets containing a wandering nontrivial continuum (and having only repelling periodic points).

This corollary contrasts with the following result of Levin: for polynomials of the form $z^{l} + c$, whose periodic points are all repelling and which have a connected Julia set, the local connectedness of the Julia set is equivalent to the nonexistence of wandering continua (see [Le]).

The following theorem strengthens the dictionary between rational maps and Kleinian groups established by D. Sullivan (see [McM2]). Indeed, every known example of finitely generated Kleinian group possesses a locally connected limit set (if it is connected). So, one conjectures that the Julia set of a genuine cubic Newton map is always locally connected. The most complete (but also more technical) result we obtain on this question is given in Proposition 8.3.

THEOREM 4. A genuine cubic Newton map, without Siegel points, has a locally connected Julia set provided the orbit of the nonfixed critical point does not accumulate on the boundary of any invariant immediate basin of attraction.

A Siegel point of a rational map R is a periodic point in the neighborhood of which R is conjugated to an irrational rotation (linearizable); the maximal domain of linearization is called a Siegel disc.

A. Douady has conjectured that for any rational map, whenever it is linearizable near a fixed point of multiplier $\lambda = e^{2i\pi\alpha}$ then α has to be a Brjuno number (see [D]). It is also conjectured that the nonlinearizability is related to the presence of small cycles (see [PM2]). In our setting we have:

THEOREM 5. If N possesses a periodic point x of multiplier $\lambda = e^{2i\pi\alpha}$ with $\alpha \in \mathbf{R}$ then

¹See Definition 6.1.

- (1) N is renormalizable at x_0 ;
- (2) $\alpha \in \mathcal{B}$ if and only if N is linearizable near x;
- (3) if $\alpha \notin (\mathcal{B} \cup \mathbf{Q})$ there exist periodic cycles $(\neq x)$ in any neighborhood of x.

An irrational α of convergents p_n/q_n (rational approximations obtained by the continued fraction development) is a *Brjuno* number, i.e. $\alpha \in \mathcal{B}$, if $\sum_{n=1}^{\infty} (\log q_{n+1})/q_n$ is finite.

All the results above are consequences of the following fundamental brick:

THEOREM 6. For every genuine cubic Newton map, without Siegel points, the connected components of the Fatou set are Jordan domains.

Theorem 6 implies Theorem 4 and, subsequently, Theorem 1, Theorem 2 and their corollaries. The proof of Theorem 1 and Theorem 2 is based on the fact that N is renormalizable at x_0 (from Theorem 5) and that the "small" Julia set touches the boundary of the basin of attraction at exactly one point which is repelling. Theorem 5 follows from the existence of puzzles with nondegenerate annuli around the critical point.

The proofs use intensively the technique of puzzles introduced by Branner and Hubbard [B-H] and developed by Yoccoz. The basic idea is to construct invariant connected graphs that divide the Julia set into connected subsets, and then to show that the iterated inverse images of these subsets shrink to points. In the case of quadratic polynomials, the graphs constructed by Yoccoz are closures of a finite number of arcs in the unbounded component of the Fatou set. In our case, instead, some edges of the graphs will have to visit infinitely many components of the basins of attraction.

The paper is organized as follows:

Section 2 gives a rough dynamical description of cubic Newton maps and then reviews the results of Yoccoz on puzzles in the context of rational-like maps.

Section 3 is devoted to finding "cut rays" which will serve as basic bricks in the construction of the puzzles.

Section 4 introduces the "articulated rays" (the main new tool in this paper) which provide nice access to points of the Julia set; periodic articulated rays are also constructed.

Section 5 yields candidates for puzzles.

Section 6 studies the renormalizations of N via puzzle pieces and gives the proof of Theorem 5.

Section 7 gives the proof of Theorem 6 with a new strategy for the renormalizable case (case 2 of Theorem 2.15).

P. ROESCH

Section 8 describes all the known cases where the Julia set is locally connected.

Acknowledgments. I would like to thank Tan Lei for her constant support and for inspiring discussions during the preparation of this work. I would also like to thank the referees for many pertinent comments.

2. Newton maps and Yoccoz puzzles

2.1. Newton maps. When one considers the dynamical system induced by a rational map f on the Riemann sphere $\widehat{\mathbf{C}}$, the interesting object is the Julia set J(f), which is the minimal compact subset, invariant by f and f^{-1} , and containing at least three points. The topological properties of J(f) give information on the dynamics of f (see [M1], [D-H1]). In particular, because of Caratheodory's Theorem on extension of conformal representations, local connectivity deserves special attention (see Remark 2.4). The difficulty of the problem depends on the behavior of critical points, and subsequently on their number.

A "simple case", usually called the geometrically finite case, occurs when the post-critical set P(f), i.e. the closure of the critical orbits of f, meets J(f)in a finite set. In this case, the Julia set is locally connected provided it is connected (see [D-H1, M2, T-Y]). To catch more interesting critical behaviors with still a tame complexity, we consider rational maps with exactly one "free" critical point. More precisely, we suppose that all critical points are simple (i.e. of local degree two) and that all are fixed (by f) except one.

LEMMA 2.1. Let $f: \widehat{\mathbf{C}} \to \widehat{\mathbf{C}}$ be a rational map of degree d having all its critical points simple, and fixed except one. Then $d \leq 3$ and f is analytically conjugate to either a quadratic polynomial or to a cubic Newton map, i.e., a map of the form

$$N(z) = z - \frac{P(z)}{P'(z)}$$
 where P is a polynomial of degree 3 with distinct roots.

Proof. A rational map of degree d possesses 2d-2 critical points and d+1 fixed points. Since the critical points are simple and only one is not fixed, there are 2d-3 distinct fixed points so that $d \leq 4$. If d = 4, every fixed point is critical which contradicts the holomorphic fixed point formula (see [M1]).

We now determine f up to conjugation by a Möbius transformation. If d = 2, we can assume that the fixed critical point is at infinity, so that f is a quadratic polynomial. Now, if d = 3 and if there are three distinct fixed critical points b_1, b_2 and b_3 (the labelling will be fixed in Notation 2.7), we can assume that the fourth fixed point of f is at infinity. We write f as follows

$$f(z) = z - \frac{P(z)}{Q(z)} = \frac{zQ(z) - P(z)}{Q(z)},$$

where P and Q are relatively prime polynomials. Since f fixes infinity, deg(Q) < deg $(zQ - P) \leq 3$ so Q has degree at most 2 and P at most 3. Moreover, $f(b_i) = b_i$, so $P(b_i) = 0$ for $i \in \{1, 2, 3\}$. Finally the condition $f'(b_i) = 0$ implies that $P'(b_i)/Q(b_i) = 1$ so Q = P' for degree reasons. Thus, f is a cubic Newton map.

The dynamics of a cubic Newton map N can be described as follows. One can always assume (up to affine conjugacy) that N is associated to a polynomial of the form $P(z) = z^3 + pz + 1$ with $p \in \mathbf{C}$; i.e.,

$$N(z) = N_P(z) = z - \frac{P(z)}{P'(z)} = \frac{2z^3 - 1}{3z^2 + p}$$

It is a degree 3 rational mapping, fixing ∞ which is repelling.² It has four critical points, one at each of the roots b_i of the polynomial P, which is a superattracting fixed point of N, and one called x_0 at 0 (which is the root of P''). Each of the roots b_i has a basin of attraction $\widetilde{B}_i = \left\{ x \in \widehat{\mathbf{C}} \mid N^n(x) \xrightarrow[n \to \infty]{} b_i \right\}$. We denote by B_i the *immediate basin* of b_i , i.e., the connected component of \widetilde{B}_i containing b_i .

The Julia set J(N) is always connected, according to results of Shishikura [Sh] (see also [T1, He]). Hence, the connected components of the *Fatou* set $F(N) = \widehat{\mathbf{C}} \setminus J(N)$ are simply connected. They are of two types, the components of the basins \widetilde{B}_i and additional (possibly empty) components due to the presence of the fourth (free) critical point x_0 . Very special situations appear when the different kinds of Fatou components mix, i.e., when x_0 belongs to some component of \widetilde{B}_i . In this case N is geometrically finite and therefore J(N) is locally connected.

The following two remarks describe in detail two particular cases where N is geometrically finite. We get rid of them through Assumption 1 for the discussion afterward.

Remark 2.2. If the critical point x_0 belongs to B_1 , B_2 or B_3 , N is quasiconformally conjugated in a neighborhood of the Julia set to the cubic polynomial $Q(z) = z^3 + \frac{3}{2}z$ (which is geometrically finite).

Proof. Assume that x_0 belongs to B_j for some $j \in \{1, 2, 3\}$. By a classical surgery process (see [CG, Ch. 6]) one conjugates N, in a neighborhood of the Julia set, to a rational map f of degree 3 possessing three fixed critical points, two of multiplicity 1 and one of multiplicity 2 (corresponding to b_j collapsed with x_0). Since the critical point of multiplicity 2, it is also backward invariant, f is conjugated (by a Möbius transformation) to a polynomial with two fixed

²A point x of period p is repelling, attracting or parabolic, respectively, if $|(f^p)'(x)| > 1$, $|(f^p)'(x)| < 1$ or $(f^p)'(x) = e^{2i\pi\theta}, \ \theta \in \mathbf{Q}/\mathbf{Z}$.

critical points in **C** of multiplicity 1. Now, up to affine conjugacy, one can assume that the polynomial is monic and centered. Hence it can be written $P(z) = z^3 + az$. Since the two finite critical points are fixed $a = \frac{3}{2}$.

Remark 2.3. If the critical point x_0 is mapped to ∞ (i.e. $N(x_0) = \infty$), N is conformally conjugated to the Newton map N_S of the polynomial $S(z) = z^3 - 1$. It is called the "symmetric case" because N_S is invariant by $z \to e^{2i\pi/3}z$.

Proof. Let P denote the polynomial associated to N (i.e. $N = N_p$). Up to affine conjugacy one can assume that $P(z) = az^3 - a$. Indeed, using a translation one can assume that $x_0 = 0$, so that $N(0) = \infty$ and $P(z) = az^3 + b$; now, using a dilatation one can assume that $P(z) = az^3 - a$. In particular, $N_P = N_S$ where $S(z) = z^3 - 1$.

The cubic polynomial $Q(z) = z^3 + \frac{3}{2}z$ is studied in [F, Ro] and so throughout the paper we will only work with genuine Newton maps. We take away the symmetric case just for technical reasons but we will dare some comments about it.

Assumption 1. From now on, we assume that the critical point x_0 does not belong to the immediate basins B_i , $i \in \{1, 2, 3\}$ and that $N(x_0) \neq \infty$.

Then the classical Böttcher Theorem [B] provides a unique conformal representation $\phi_j : \mathbf{D} \to B_j$ — where **D** is the open unit disc — which conjugates N to the map $z \mapsto z^2$. Each ϕ_j induces polar coordinates on B_j . The ray of angle $t \in \mathbf{R}/\mathbf{Z}$, denoted by $R_j(t)$, is the arc $\phi_j([0, 1]e^{2i\pi t})$ and the equipotential of level v > 0 is defined as $E_j(v) = \phi_j(e^{-v}\mathbf{S}^1)$. These objects can be lifted to the other components of \widetilde{B}_j as long as they don't contain the critical point x_0 (see Notation 2.9 below).

Remark 2.4. Once Theorem 6 has been proved, the local connectedness of ∂B_j will give that the map ϕ_j extends continuously to $\overline{\mathbf{D}}$ so that ∂B_j is a curve, by Carathéodory's Theorem. Then Lemma 3.8 will give the injectivity of this extension. Hence ∂B_j is a Jordan curve on which $N \mid_{\partial B_j}$ will be conjugated to the map $\theta \mapsto 2\theta$ on \mathbf{S}^1 (by the extension ϕ_j^{-1}).

Recall that a ray $R_j(t)$ is said to *converge* (or to *land*) if the quantity $\phi_j(e^{-v+2i\pi t})$ has a limit when v tends to 0. Douady, Hubbard and Sullivan proved that for every rational angle t, the ray $R_j(t)$ converges to an eventually periodic point in ∂B_j , which is repelling or parabolic, with period dividing p where t is written in the form $t = r/(2^m(2^p - 1))$ (see [D-H1, M1]).

The following lemma shows a dis-symmetry of the Julia set of N:

LEMMA 2.5. The three rays $R_1(0)$, $R_2(0)$ and $R_3(0)$ converge to ∞ . On the contrary, only two of the rays of angle 1/2 do converge to the same preimage



Figure 1: On the left B_1, B_2, B_3 and on the right the rays $R_i(0)$.

of ∞ (when $N(x_0) \neq \infty$, i.e. in the general case). We denote this preimage by ξ . The other ray of angle 1/2 converges to $-\xi$.

Note that in the particular case where $N(x_0) = \infty$ (avoided by assumption 1), all the rays of angle 1/2 converge to the preimage of ∞ which is x_0 .

The following fact will be useful for the proof of Lemma 2.5 and several times later also:

TRIVIAL FACT 2.6. If two distinct rays converge to the same point p and have the same image under N then p is a critical point of N.

Indeed, N is not injective near p.

Proof of Lemma 2.5. Each ray $R_j(0)$ is fixed by N, and so it converges to a fixed point on ∂B_j that can only be ∞ . Each ray of angle 1/2 lands at a preimage of ∞ , distinct from ∞ by the above remark.

If $N(x_0) = \infty$, the map near x_0 is a double cover over a neighborhood of ∞ , so that there are two preimages of each ray $R_i(0)$ near x_0 . Hence each ray of angle 1/2 lands at x_0 .

If $N(x_0) \neq \infty$, we assume by contradiction that the three rays $R_1(1/2)$, $R_2(1/2)$, $R_3(1/2)$ land at the same inverse image of ∞ , denoted by x. Their cyclic order at x is different from the cyclic order of their images $R_1(0)$, $R_2(0)$, $R_3(0)$ at ∞ . This contradicts the conformality of N at x.

We will use the following convention for labelling basins (see Figure 1):

Notation 2.7. In the general case, i.e. if $N(x_0) \neq \infty$, the basins are labelled B_1, B_2, B_3 in such a way that:

- the two rays of angle 1/2 landing at the same preimage ξ of $N^{-1}(\infty)$ are in B_1 and B_2 ;
- the rays of angle 0 of B_1, B_2, B_3 land at ∞ in positive cyclic order.

In the symmetric case, i.e. if $N(x_0) = \infty$, the labelling B_1 , B_2 , B_3 of the basins is without importance because of the symmetry.

Remark 2.8. Under assumption 1, the restriction $N: B_j \to B_j$ is a degree two ramified covering. So $N^{-1}(B_j)$ consists of B_j and another connected component which is a topological disc since N can only induce a nonramified covering of degree one on it.

Notation 2.9. We denote by B'_j the connected component of $N^{-1}(B_j)$ disjoint from B_j .

If U is a disc such that $N^p: U \to B_j$ is a homeomorphism, we denote by $R_U(t)$ (resp. $E_U(v)$) the ray of angle t in U, i.e., $(N^p_{|U})^{-1}(R_j(t))$, (resp. the equipotential of level v in U, i.e., $(N^p_{|U})^{-1}(E_j(v))$). More quickly for B'_j we adopt the notation $R'_i(t)$ and $E'_i(v)$ respectively.

Remark 2.10. For j = 1, 2 the rays $R'_j(0)$ and $R_3(1/2)$ land at the same inverse image ξ' (possibly x_0) of ∞ . This follows from Lemma 2.5 and the fact that $R'_k(0)$ and $R_k(1/2)$ (or $R_k(0)$) k = 1, 2, 3, cannot land at a common noncritical point because they are all mapped to $R_k(0)$ (trivial fact).

2.2. *Yoccoz Puzzles.* We now describe the technique of puzzles in the convenient framework of rational-like maps.

Definition 2.11. Let X, X' be connected open subsets of $\widehat{\mathbf{C}}$ with finitely many smooth boundary components and such that $\overline{X'} \subset X$. A holomorphic map $f: X' \to X$ is called a *rational-like map* (resp. a *simple rational-like map*) if it is proper and has finitely many critical points in X (resp. a single critical point with multiplicity one). We denote by degree(f) the topological degree of f and by $K(f) = \{x \in X' \mid \forall n \geq 0, \quad f^n(x) \in X'\} = \bigcap_{n \geq 0} f^{-n}(X)$, the associated filled Julia set.

For simply connected domains this is the standard definition of *polynomial-like maps* ([D-H2]).

Example 2.12. Any genuine cubic Newton map N induces a simple rationallike map $N: X' \to X$ (see Figure 2) for any potential v, where

$$X = \widehat{\mathbf{C}} \setminus \bigcup_{i=1,2,3} \phi_i(e^{-v}\overline{\mathbf{D}}) \text{ and } X' = N^{-1}(X).$$

134



Figure 2: N viewed as a rational like map from X' on the left to X on the right.

Definition 2.13. A finite, connected, graph Γ in \overline{X} is called a *puzzle* if it satisfies the conditions: $\partial X \subset \Gamma$, $f(\Gamma \cap X') \subset \Gamma$, and the orbit of each critical point avoids Γ .

The puzzle pieces of depth n are the connected components of $f^{-n}(X \setminus \Gamma)$ and the one containing a point x is denoted by $P_n(x)$. If a point x is contained in a puzzle piece at each depth, i.e. if the orbit of x avoids Γ , the sequence $P_n(x)$ is decreasing and the *impression* of x is defined to be the set $\text{Imp}(x) = \cap \overline{P}_n(x)$. A puzzle is said to be k-periodic at x if $f^k(P_{n+k}(x)) = P_n(x)$ for any sufficiently large n.

Every difference set $A_n = P_n \setminus \overline{P}_{n+1}$ between two nested puzzle pieces P_n, P_{n+1} of consecutive depths is called a *puzzle annulus of depth n*. This "annulus" actually degenerates to a disc if ∂P_n meets ∂P_{n+1} . A point $x \in K(f)$ is said to be *surrounded* at depth *n* if $f^{n+1}(x) \notin \Gamma$ and $P_n(x) \setminus \overline{P}_{n+1}(x)$ is a nondegenerate (i.e. genuine) annulus. A point is *infinitely surrounded* if it is surrounded at infinitely many depths.

Remark 2.14. One can also consider closed annuli $\overline{P}_n \setminus P_{n+1}$. When such closed annuli degenerate, they still surround the points of P_{n+1} since the closed annuli still have the homotopy type of the circle.

The following theorem is a rewording (see [Ro] for the proof) of Yoccoz' Theorem (see [Hu, M2, Y]). The introduction of different puzzles Γ^i is natural since it is difficult to surround every point of the Julia set with just one puzzle.

P. ROESCH

THEOREM 2.15. Let $f: X' \to X$ be a simple rational-like map, C a subset of K(f) such that $f(C) \subset C$ and let $\Gamma^0, \ldots, \Gamma^r$ be puzzles all surrounding the unique critical point x_0 . If every point of C is surrounded by some Γ^j at a uniformly bounded depth, then for every point $x \in C$ there exists $i \in \{0, \ldots, r\}$ such that x is infinitely surrounded by Γ^i and the following dichotomy holds:

- 1) If the puzzle Γ^i is not periodic at x_0 then the impression Imp^i for Γ^i satisfies $\text{Imp}^i(x) = \{x\};$
- 2) If the puzzle Γ^i is k-periodic at x_0 then $f^k \colon P^i_{n+k}(x_0) \to P^i_n(x_0)$ defines a polynomial-like map of degree 2 for some large n and its filled Julia set is $\operatorname{Imp}^i(x_0)$; moreover, depending on whether the orbit of x meets $\operatorname{Imp}^i(x_0)$ or not, $\operatorname{Imp}^i(x)$ is either a conformal copy of $\operatorname{Imp}^i(x_0)$ or $\{x\}$.

We will apply Theorem 2.15 to $C = \partial B_i$ with the puzzles $\Gamma^0, \ldots, \Gamma^r$ constructed in Section 5 and 7, consisting of rays, articulated rays and equipotentials. For any point $x \in C$ such that $\text{Imp}(x) = \{x\}$, the sets $\overline{P}_n(x) \cap C$ form a basis of neighborhoods of x in C so that, if they are all connected, C is locally connected at x.

A useful trick for finding points surrounded by the puzzles is contained in the following remark. It occurs in the proof of our version of Yoccoz' Theorem (Theorem 2.15) and will be crucial in Proposition 5.4 (see [Ro]).

Remark 2.16. If an iterated image $f^k(x)$ of a point x is surrounded by a nondegenerate annulus $P_0(f^k(x)) \setminus \overline{P_1(f^k(x))}$ (for some puzzle Γ) then x itself is surrounded by Γ at depth k and, more precisely, by the nondegenerated annulus $P_k(x) \setminus \overline{P_{k+1}(x)}$. This follows from the fact that f^k induces a proper map $P_k(x) \to P_0(f^k(x))$ taking $P_{k+1}(x)$ to $P_1(f^k(x))$.

3. Cut angles and localization of x_0

Here and below, we identify \mathbf{S}^1 with \mathbf{R}/\mathbf{Z} , so that -t and 1-t coincide. We write $t_1 \leq t \leq t_2$ to mean that t_1, t, t_2 are in trigonometric order. To compare only two angles, we use their representatives in (0, 1].

A preliminary step for proving that ∂B_i is locally connected is to find small connected subsets in ∂B_i . The following basic examples will be used throughout the paper.

Example 3.1. Let $Q \subset \overline{B}_k$ denote the closure of the sector between two converging rays $R(t_1), R(t_2)$, namely,

$$Q = \overline{\{\phi_k(r \, e^{2i\pi t}), \ r \in [0, 1), \ t_1 \le t \le t_2\}}.$$

Then $Q \cap \partial B_k$ is a connected subset of ∂B_k . Indeed it is the decreasing intersection of the compact connected sets \overline{S}_n , where

$$S_n = \left\{ \phi_k(r e^{2i\pi t}), \ r \in [1 - 1/n, 1), \ t_1 \le t \le t_2 \right\} \text{ for } n \ge 1.$$

To construct puzzles, we will use rays of different basins converging to the same point in the Julia set:

Definition 3.2. We say that a ray $R_i(t) \subset B_i$ is a *cut ray* if it converges to the landing point of another ray $R_j(t')$ where $j \neq i$. The angle $t \in S^1$ of such a ray is called a *cut angle* in B_i .

The goal of this section is to determine the set of cut angles. Most properties we establish are proved in [He] and [T1] but under the assumption that J(N) is locally connected. For this reason, detailed arguments are provided here, though most of them are identical. The essential difference resides in Proposition 3.12 and Lemma 3.15.

3.1. Rays landing at the same point.

PROPOSITION 3.3. The only cut angle in B_3 is 0. Furthermore, t is a cut angle in B_1 if and only if 1-t is a cut angle in B_2 and, if so, the corresponding cut rays $R_1(t)$ and $R_2(1-t)$ land at the same point.

Let G denote the set of angles $t \in \mathbf{S}^1$ such that $R_1(t)$ and $R_2(1-t)$ land at the same point. The above proposition asserts that G is the set of cut angles in B_1 and that the set of cut angles in B_2 is $\{1 - t, t \in G\}$.

Remark 3.4. G contains 0 and 1/2 by Lemma 2.5.

LEMMA 3.5. For $i \in \{1, 2, 3\}$, the only ray in B_i converging to ∞ is $R_i(0)$.

Proof. Assume that $R_i(\theta)$ converges to ∞ for some *i* and some $\theta \in (1/4, 1/2)$. Then the rays $R_i(\theta + 1/2)$, $R_i(1/2)$ and $R_i(2\theta - 1/2)$ all land at points of $N^{-1}(\infty) \setminus \{\infty\}$ (see Trivial Fact 2.6). But $0 < 2\theta - 1/2 < \theta < 1/2$ and $1/2 < 2\theta < \theta + 1/2 < 1$ for the cyclic order, so each of the three connected components of $\widehat{\mathbf{C}} \setminus \delta_i$, where $\delta_i = \overline{R}_i(0) \cup \overline{R}_i(\theta) \cup \overline{R}_i(2\theta)$, contains the landing point of exactly one of those rays. This contradicts the fact that only two preimages of ∞ lie outside of δ_i . Now, if $R_i(\theta)$ lands at ∞ with $\theta \notin (1/4, 1/2)$ and θ nondyadic, some $2^n\theta \mod 1$ is in (1/4, 1/2) and the above applies. Finally no rays of nonzero dyadic angle converge to ∞ by Trivial Fact 2.6.

LEMMA 3.6. Let $\theta \in \mathbf{S}^1$ be such that $2\theta \in G$. If $\theta \notin G$ then the rays $R_1(\theta)$ and $R'_2(1-2\theta)$ converge to the same point, as well as $R_2(1-\theta)$ and $R'_1(2\theta)$. Moreover, the two landing points are distinct. On the other hand, if $\theta \in (0, 1/2)$ belongs to G, the rays $R'_1(2\theta)$ and $R'_2(1-2\theta)$ land at the same point.

Proof. We assume that $\theta \neq 0$ and 1/2, the special case $\theta = 1/2$ follows from the definition of B_1 and B_2 . Since $2\theta \in G$, the rays $R_1(2\theta)$ and $R_2(1-2\theta)$

converge to the same point, denoted x. Hence, each of the six rays $R_1(\theta)$, $R_1(\theta + 1/2)$, $R'_1(2\theta)$, $R_2(1 - \theta)$, $R_2(1/2 - \theta)$, $R'_2(1 - 2\theta)$ lands at a preimage of x. Opposite rays $R_i(t)$ and $R_i(t + 1/2)$ are separated by the Jordan curve $\gamma = \overline{R}_1(0) \cup \overline{R}_1(1/2) \cup \overline{R}_2(0) \cup \overline{R}_2(1/2)$. Since no preimage of x is on this curve (Lemma 3.5), γ separates the six closed rays into two groups. If $x = N(x_0)$, then x has only two preimages and all the rays in each group converge to the same point. If $x \neq N(x_0)$, then x has three preimages and only one preimage of $R_1(2\theta)$ (resp. of $R_2(1 - 2\theta)$) lands at each of them (Trivial Fact 2.6). Thus if $\theta \notin G$, the only possibility is that $R_1(\theta)$ and $R'_2(1 - 2\theta)$ land at the same point, as well as $R_2(1 - \theta)$ and $R'_1(2\theta)$. In the case where $\theta \in G$ let us assume that $\theta \in (0, 1/2)$ (else $\theta + 1/2 \in (0, 1/2)$). Then $R_1(\theta + 1/2)$ and $R_2(1/2 - \theta)$ are separated from the four other rays by γ (by Remark 2.10 B'_1 and B'_2 are in the other component of $\overline{\mathbb{C}} \setminus \gamma$). Hence they converge to the same point, as well as $R'_1(2\theta)$ and $R'_2(1 - 2\theta)$.

COROLLARY 3.7. For every $n \in \mathbf{N}$, the angle $1 - 1/2^n$ belongs to G but there exists a smallest n_0 so that $1/2^{n_0}$ is not in G.

Proof. An easy induction using Lemma 3.6 shows that $R_1(1-1/2^n)$ and $R_2(1/2^n)$ land at the same point since the curve $\gamma = \overline{R}_1(0) \cup \overline{R}_1(1/2) \cup \overline{R}_2(0) \cup \overline{R}_2(1/2)$ separates \overline{B}'_2 and $\overline{R}_1(1-1/2^n)$ for $n \geq 2$.

Assume now that $1/2^n \in G$ for all $n \in \mathbb{N}$. Let D_n be the unbounded connected component of $\widehat{\mathbb{C}} \setminus (\overline{R}_1(\pm 1/2^n) \cup \overline{R}_2(\pm 1/2^n) \cup E_1(2^n) \cup E_2(2^n))$ and $A_n = D_n \setminus \overline{D}_{n+1}$. It is easy to check that N induces a homeomorphism from A_n to A_{n-1} , so that the annuli A_n have equal moduli. Therefore, $\bigcap D_n = \{\infty\}$ (see [A]) which contradicts the fact that $B_3 \subset D_n$.

LEMMA 3.8. Two different rays in the same basin B_i cannot converge to the same point.

Proof. Given $\theta, \theta' \in [0, 1)$ with $\theta < \theta'$, there is some $n \in \mathbf{N}$ such that $2^n \theta$ and $2^n \theta'$ are distinct and belong to [0, 1/2] and [1/2, 1] respectively. If $R_i(\theta)$ and $R_i(\theta')$ land at the same point, then $R_i(2^n\theta)$ and $R_i(2^n\theta')$ converge to ∞ or $\zeta \in f^{-1}(\infty)$. Indeed, $R_i(2^n\theta)$ and $R_i(2^n\theta')$ lie in different components of $\widehat{\mathbf{C}} \setminus \gamma_i$ where $\gamma_1 = \gamma_2 = \bigcup_{j=1,2} \overline{R_j}(0) \cup \overline{R_j}(1/2)$. Using the integer n_0 given by Corollary 3.7, we have

$$\gamma_3 = \overline{R}_1(0) \cup \overline{R}_1(1/2^{n_0-1}) \cup \overline{R}_2'(1-1/2^{n_0}) \cup \overline{R}_2'(0) \cup \overline{R}_3(1/2) \cup \overline{R}_3(0)$$

(γ_3 is a curve by Lemma 3.6). By the Trivial Fact 2.6, a nonzero dyadic angle cannot land at ∞ , and so, after iterations, the result follows from Lemma 3.5.



Figure 3: On the left, the curve γ_3 used in Lemma 3.8. On the right Head's angle (Definition 3.11)

COROLLARY 3.9. Fix some $i \in \{1, 2, 3\}$ and $m \ge 0$. Let U, V be connected components of $N^{-m}(B_i)$. If two rays $R_U(t)$ and $R_V(t')$ land at the same point then either U = V and t = t', or the landing point is an iterated preimage of the critical point x_0 .

Proof. The rays $N^m(R_U(t))$ and $N^m(R_V(t'))$ both lie in B_i and coincide because they land at the same point (Lemma 3.8). Let $k \leq m$ be the smallest integer such that $N^k(R_U(t))$ is equal to $N^k(R_V(t'))$. If $t \neq t'$ or $U \neq V$ then $k \geq 1$ and the common landing point of $N^{k-1}(R_U(t))$ and $N^{k-1}(R_V(t'))$ is x_0 by the Trivial Fact 2.6.

Remark 3.10. The proofs of Lemmas 3.5, 3.6, 3.8 and the corollaries 3.7, 3.9 still work in the case where $N(x_0) = \infty$ (with $n_0 = 2$), but not the proof of Proposition 3.3.

Proof of Proposition 3.3. The idea is to study the position of the rays relative to the curve $\gamma = \overline{R}_1(0) \cup \overline{R}_2(0) \cup \overline{R}_1(1/2) \cup \overline{R}_2(1/2)$.

Assume first that $R_1(\theta)$ and $R_2(\theta')$ land at the same point x. As long as $N^n(x)$ avoids γ , the iterated images of the two rays lie in the same connected component of $\widehat{\mathbf{C}} \setminus \gamma$. Hence θ' and $1 - \theta$ have the same dyadic expansion. If the point $N^n(x)$ falls in γ for some n then $2^n\theta = 2^n\theta' = 0$ or 1/2 (Lemma 3.8) and equality $\theta' = 1 - \theta$ follows since all previous terms in the dyadic expansion coincide.

Assume now that the rays $R_3(\theta)$ and $R_1(\theta')$ land at the same point, with $(\theta, \theta') \neq (0, 0)$. If $2^n \theta' = 1/2$ for some $n \geq 0$, the rays $R_3(2^n \theta)$, $R_2(1/2)$,

 $R_1(1/2)$ land at the same preimage ξ of ∞ in a cyclic order different from that of their images $R_3(2^{n+1}\theta)$, $R_2(0)$, $R_1(0)$ — observe that $R_3(2^n\theta)$ and $R_3(2^{n+1}\theta)$ are in the same connected component of $\widehat{\mathbf{C}} \setminus \gamma$. This contradicts assumption 1 so there exists $n \geq 0$ such that $2^n\theta' \mod 1 \in (1/2, 1)$. The ray $R_1(2^n\theta')$ is separated from B_3 by γ (look at the cyclic order in B_1) and so has to land at ξ or ∞ . But θ' would be dyadic (Lemma 3.8). The same argument works for B_2 instead of B_1 .

3.2. Characterization of cut angles and localization of x_0 .

Definition 3.11. Let $\alpha = \alpha(N)$ be the infimum of G for the order obtained by identifying $\mathbf{S}^1 = \mathbf{R}/\mathbf{Z}$ with (0, 1]. The angle α is called *Head's angle* of N (see [He], [T1]).

If an angle θ belongs to G then $2^n \theta$ is in $[\alpha, 1]$ for every $n \ge 0$ because $2^n \theta \in G$. This property characterizes rational cut angles:

PROPOSITION 3.12. Let $\theta \in \mathbf{Q}/\mathbf{Z}$. If $2^n\theta$ belongs to $(\alpha, 1]$ for all $n \ge 0$ then θ is a cut angle in B_1 , i.e. $\theta \in G$.

LEMMA 3.13. If $\theta \in (\alpha, 1]$ and $2\theta \in G$ then $\theta \in G$.

Proof. Let $t < \theta$ be a point in G. The curve $\overline{R}_1(0) \cup \overline{R}_2(0) \cup \overline{R}_1(t) \cup \overline{R}_2(1-t)$ separates $R_1(\theta)$ and B'_2 , so θ is in G by Lemma 3.6.

To prove Proposition 3.12, we will construct open discs $U \subset V$ containing the landing points x_1, x_2 of $R_1(\theta)$, $R_2(1 - \theta)$ and such that N^k induces a homeomorphism $U \to V$, where θ is of period k under doubling. We will then conclude that $x_1 = x_2$ by the Schwarz' Lemma. The following localisation Lemma will be used to leave the critical point x_0 out of U:

LEMMA 3.14. Let $\eta \in G$ and let $\gamma(\eta) = \overline{R}_1(0) \cup \overline{R}_1(\eta) \cup \overline{R}_2(0) \cup \overline{R}_2(1-\eta)$. The critical points x_0 and $\phi_1(]0, \eta[)$ are in the same connected component of $\widehat{\mathbf{C}} \setminus \gamma(\eta)$. More precisely, given $\kappa \in \mathbf{S}^1 \setminus G$ such that $2\kappa \in G$, the set $\gamma(\eta, \kappa) = \overline{R}_1(\eta) \cup \overline{R}_2(-\eta) \cup \overline{R}_1(\kappa) \cup \overline{R}_2(-\kappa) \cup \overline{R}_1'(2\eta) \cup \overline{R}_2'(-2\eta) \cup \overline{R}_1'(2\kappa) \cup \overline{R}_2'(-2\kappa)$ separates x_0 from ∞ .

Proof. The first assertion readily follows from the second one. On the other hand, we may assume $\eta \leq 1/2$ since this just makes the bounded connected component of $\widehat{\mathbf{C}} \setminus \gamma(\eta, \kappa)$ smaller. By Lemma 3.6, the curve $\gamma = \gamma(\eta, \kappa)$ and its image $N(\gamma) = \overline{R}_1(2\eta) \cup \overline{R}_2(1-2\eta) \cup \overline{R}_1(2\kappa) \cup \overline{R}_2(1-2\kappa)$ are Jordan curves. Therefore, the bounded component B of $\widehat{\mathbf{C}} \setminus \gamma$ is a disc and its image N(B)is also a disc (a component of $\widehat{\mathbf{C}} \setminus N(\gamma)$) because $\eta, \kappa \leq 1/2$ and the curve $N^{-1}(N(\gamma)) = \gamma \cup \gamma'$ with $\gamma' = \overline{R}_1(\eta + 1/2) \cup \overline{R}_2(1/2 - \eta) \cup \overline{R}_1(\kappa + 1/2) \cup \overline{R}_2(1/2 - \kappa)$ is disjoint from B. Finally $x_0 \in B$ since $N_{|_{\partial B}}$ has degree two. \Box



Figure 4: Illustration of Lemma 3.14: on the left, the curve $\gamma(\eta)$, on the right the localization of the critical point x_0 in the bounded component of $\overline{\mathbf{C}} \setminus \gamma(\eta, \kappa)$.

Proof of Proposition 3.12. The angle $\theta \in \mathbf{Q}/\mathbf{Z}$ is (eventually) periodic by $t \to 2t$ since it can be written in the form $r/(2^p(2^k - 1))$. Therefore, using Lemma 3.13, we now assume that $2^n\theta$ belongs to $(\alpha, 1)$ for every $n \ge 0$ and that θ is periodic.

1. We first claim that θ is accumulated on both sides by elements of G. Indeed, given $\epsilon > 0$, let n be the first integer such that the interval $2^n[\theta, \theta + \epsilon)$ intersects $(0, \alpha)$. Since $2^n \theta$ lies in $(\alpha, 1)$ there exists $t' \in (\theta, \theta + \epsilon)$ so that $2^n t' = 1$, and hence t' belongs to G (Lemma 3.13). The same argument yields a point $t \in (\theta - \epsilon, \theta) \cap G$.

2. Let k be the period of θ and denote by x_1, x_2 the respective landing points of $R_1(\theta), R_2(1-\theta)$; these points are periodic and their period divides k. Let $t, t' \in G$ be angles surrounding θ (as in point 1.) and let U be the bounded connected component of $\mathbb{C} \setminus \overline{R}_1(t) \cup \overline{R}_1(t') \cup \overline{R}_2(1-t) \cup \overline{R}_2(1-t')$. The disc U contains x_1 and x_2 . Moreover for ϵ small enough, $V = N^k(U)$ is also a disc and covers U since multiplication by 2^k is expanding and fixes $\theta = 2^k \theta$. To see that $N^k : U \to V$ is a (conformal) homeomorphism, we need to check that x_0 is in no $N^i(U)$ for $0 \le i < k$. Let $\eta \in G$ be such that $\eta < \inf\{2^i\theta \mod 1 \mid i \le k\}$. Then for ϵ small enough $2^i(t, t') \subset (\eta, 1)$ for $i \le k$, so that $N^i(U)$ is not in the component of $\widehat{\mathbb{C}} \setminus \overline{R}_1(0) \cup \overline{R}_1(\eta) \cup \overline{R}_2(0) \cup \overline{R}_2(1-\eta)$ containing x_0 . Injectivity now follows from Lemma 3.14 and $\left(N_{|_U}^k\right)^{-1}$ has only one fixed point so that $x_1 = x_2$.

Remark. Proposition 3.12 does not show that the Head angle α (assuming it is rational) belongs to G (the above argument fails). This will however follow

from the local connectivity of ∂B_i , $i \in \{1, 2\}$ (by continuity of the extended Böttcher map).

3.3. Examples of periodic cut angles.

LEMMA 3.15. The Head angle α belongs to (0, 1/2) (under assumption 1).

Proof. Assuming $\alpha = 0$, Proposition 3.12 implies that $1/2^n$ is in G for every $n \geq 0$, which contradicts Corollary 3.7. On the other hand, $\alpha \leq 1/2$ since $1/2 \in G$. Suppose now that $\alpha = 1/2$. The critical point x_0 belongs to the bounded connected component U_n of $\widehat{\mathbf{C}} \setminus \gamma_n$, where $\gamma_n = \gamma(1/2, 1/2 - 1/2^n)$ (see Lemma 3.14). Hence $N(x_0)$ belongs to $\bigcap_{n \in \mathbb{N}} V_n$ where V_n is the unbounded connected component of $N(U_n) \setminus (E_1(2^n) \cup E_2(2^n))$. This is impossible since, as we will now show, $\bigcap_{n \in \mathbb{N}} V_n = \emptyset$.

Let ψ be a homeomorphism $\overline{V}_1 \to \overline{\mathbf{D}} \cap \{z = x + iy \mid y \ge 0\}$ which is conformal on V_1 and maps $(\overline{R}_1(0) \cup \overline{R}_2(0)) \cap \partial V_1$ to [-1,1]. The map Ninduces a homeomorphism from \overline{V}_2 onto \overline{V}_1 and we set $g = \psi \circ f \circ \psi^{-1}$ where $f : \overline{V}_1 \to \overline{V}_2$ denotes the inverse branch of N. Since g preserves [-1,1], it extends by reflection to a holomorphic map $\tilde{g} : \mathbf{D} \to \mathbf{D}$. Now $\tilde{g}(\mathbf{D})$ has compact closure in \mathbf{D} , so the intersection of the domains $\tilde{g}^n(\overline{\mathbf{D}})$ is reduced to a point. Similarly, $\cap g^n(\psi(\overline{V}_1))$ is a point and hence $\cap \overline{V}_n = \{\infty\}$. \Box

COROLLARY 3.16. For n large enough (depending on α), the angle $1 - \frac{1}{2^n-1}$ belongs to G.

Proof. If $\theta_n = 1 - \frac{1}{2^n - 1}$, then $2^i \theta_n = 1 - \frac{2^i}{2^n - 1}$ is in (1/2, 1) for i < n - 1 and $2^{n-1}\theta_n$ equals $\frac{1}{2} - \frac{1}{2}\frac{1}{2^n - 1}$ and so is in $(\alpha, 1)$ for n large enough. Hence, by Proposition 3.12, θ_n belongs to G.

To construct articulated rays in Section 4 (Proposition 4.3), we will need the following lemma:

LEMMA 3.17. For any $\zeta \in G$ with $\zeta < 2\alpha$, there exists $\theta \in G$ dyadic such that $\zeta \leq \theta < 2\alpha$.

Proof. Let $\beta = \sup\{t \notin G \mid 2t \in G\}$. Note that $\beta \leq \alpha \leq 2\beta$. Moreover, $t > \beta \implies t \in G$ or $2t \notin G$.

1. We prove that if $\beta < \alpha$ then β is dyadic and that $\theta = 2\beta$ works: The interval $2^{j}(\beta, \alpha)$ does not intersect G provided $2^{i}(\beta, \alpha) \subset (\beta, 1)$ for $1 \leq i \leq j-1$. The first interval $2(\beta, \alpha)$ always lies in $[\alpha, 1)$ because $\alpha < 1/2$ and does not intersect G by definition of β . The rest follows by induction. Hence, the first interval $2^{n}(\beta, \alpha)$ that meets $(0, \beta)$ is disjoint from G, and so is included in $[0, \alpha]$. Assume that $2^{n}\beta \neq 0 \mod 1$. By definition of β , there exists a sequence $\beta_{k} < \beta$ converging to β such that $2\beta_{k} \in G$ and so $2^{n}\beta_{k} \in G$. This contradicts

the fact that for k large enough $0 < 2^n \beta_k < 2^n \beta \leq \alpha$. Finally, $\theta = 2\beta \in G$ by Proposition 3.12 and $2\beta \geq \zeta$ since $\zeta \in G$ and $2(\beta, \alpha)$ does not intersect G.

2. Assume now that $\beta = \alpha$ and let $\beta_k \in (0, \beta)$ be angles converging to β such that $2\beta_k \in G$. Let also $n \ge 1$ be the smallest integer such that $2^n(\beta_k, \beta)$ intersects $(0, \alpha)$. Then either $2^n\beta_k = 0 \mod 1$ or $1 \in (2^n\beta_k, 2^n\beta)$ (on the circle). Therefore, $G \cap (2\beta_k, 2\beta)$ contains a dyadic angle which, for k large enough, is arbitrarily close to $2\beta = 2\alpha$.

4. Articulated rays

Cut rays can be used to construct puzzles. Indeed, a typical example of such a puzzle consists of equipotentials $E_i(v)$ (for any v > 0), the rays $\overline{R}_i(0)$, with *i* in $\{1, 2, 3\}$ and some properly chosen periodic cycle of cut rays (see Section 5). However, these graphs all contain the point ∞ and therefore, even taking different cut angles, it seems difficult to surround points close to ∞ in order to apply Theorem 2.15.

The articulated rays considered in this section avoid the point ∞ . They are arcs (connecting B_1, B_2 to B_3 away from the rays $R_i(0)$) whose behavior under the dynamics is similar to that of rays. "Periodic" articulated rays will be especially useful to build puzzles. Articulated rays will also reappear in Section 7.

Definition 4.1. An articulated ray stemming from B_i with angle θ is a curve L satisfying the following properties:

- $L = \bigcup_{k \ge 0} \overline{l}_k$ where \overline{l}_k is the closure of a converging ray l_k ;
- $l_0 = R_i(\theta)$ and the depth of l_1 is greater than that of l_0 (i.e. is at least 1);
- \overline{l}_k and \overline{l}_{k+1} intersect in exactly one point;
- l_{k+2} is of depth greater than l_k .

A curve $L = \bigcup_{0 \le k \le m} \overline{l}_k$ with the same properties will be called a *finite articulated* ray. Articulated rays stemming from any preimage U of B_i are defined

similarly, just by replacement of $R_i(\theta)$ by $R_U(\theta)$ in the second item. Recall that the *depth* of a ray $R_U(\theta)$ is the smallest integer p such that

 $N^p(U) = B_j$ for some $j \in \{1, 2, 3\}$.

Each articulated ray $L = \bigcup_{k \ge 0} \overline{l}_k$ has a natural parametrization $\rho \colon [0, +\infty)$

 $\rightarrow \widehat{\mathbf{C}}$: for every $k \geq 0$ and every $t \in [0,1]$, the points $\rho(2k+t)$ and $\rho(2k+2-t)$ belong to l_{2k} and l_{2k+1} respectively, and their Böttcher coordinates have modulus t. We say that L converges to a point y if $\rho(t)$ has limit y as t goes to ∞ .



Figure 5: The articulated rays L^0, L^1, L^2 and the image $N^3(L)$ which is L union the dashed rays.

Remark 4.2. Two rays l_i, l_j of an articulated ray L lie in the same Fatou component if and only if $\{i, j\} = \{2k - 1, 2k\}$ for some $k \ge 1$. Moreover, L has no self-intersection unless it contains some iterated preimage of x_0 (Lemma 3.8 and Corollary 3.9).

Corollary 3.16 gives a lot of nondyadic angles in $G \cap (\alpha, 2\alpha)$. We use them now to construct a "3-periodic" articulated ray, meaning that $N^3(L)$ differs from L only from a finite number of rays of depth 0.

PROPOSITION 4.3. Let $\zeta \in G \cap (\alpha, 2\alpha)$ be a nondyadic rational angle and y the landing point of the ray $R_3(1/7)$. There exists a unique articulated ray L stemming from B_2 with angle $-\zeta/4$, converging to y and satisfying the 3-periodicity condition:

$$N^{3}(L) = L \cup \left(\overline{R}_{1}(\zeta) \cup \overline{R}_{2}(-\zeta) \cup \overline{R}_{1}(2\zeta) \cup \overline{R}_{2}(-2\zeta)\right).$$

The method developed below can actually be used to obtain periodic articulated rays converging to any periodic point of ∂B_3 , precisely; given a *k*-periodic point *y* on ∂B_3 , there exists a unique articulated ray *L* stemming from $B_1 \cup B_2$, landing at *y* and satisfying $L \subset N^k(L) \subset L \cup \bigcup_{i\geq 0} (R_1(2^i\zeta) \cup R_2(-2^i\zeta))$. The general argument, however, is more tricky and is unnecessary for Theorems 1 and 6, so we will not make it here. The basic idea is to trace the itinerary of the periodic point and to build the articulated ray backward.

Proof of Proposition 4.3. Let $\theta \in G$ be a dyadic angle with $\zeta \leq \theta < 2\alpha$ (Lemma 3.17 provides some) and denote by V the connected component of $\widehat{\mathbf{C}} \setminus (\overline{R}_1(0) \cup \overline{R}_2(0) \cup \overline{R}_1(\theta) \cup \overline{R}_2(-\theta) \cup \overline{R}_3(0) \cup E_3(v)) \ (v > 0)$ which does not contain the critical value $N(x_0)$ (see Lemma 3.14 and Figure 6). Since V is a disc, $N^{-1}(V)$ has three connected components and, for $i \in \{1, 2\}$, only one of them intersects both B_3 and B_i ; we denote it by U_i (see Figure 6) and proceed in four steps.



Figure 6: The domains U_1, U_2 and V.

1. Uniqueness. Since $\zeta/2 \notin G$, the curve N(L) is an articulated ray (stemming from B_2 with angle $-\zeta/2$ but $N^2(L)$ is not. Indeed, the announced form of $N^3(L)$ shows that the second ray in $N^2(L)$ lies in B'_1 or in B_1 but B'_1 is ruled out for the following reason: $\zeta > \alpha$, so that there exists $\theta' \in G \cap [\alpha, \zeta)$ and the curve $\overline{R}_1(0) \cup \overline{R}_2(0) \cup \overline{R}_1(\theta') \cup \overline{R}_2(-\theta')$ separates $\overline{R}_2(-\zeta)$ from \overline{B}'_1 (Remark 2.10 and the cyclic ordering of angles). Hence, the image $N^2(L)$ equals $\overline{R}_1(\zeta) \cup \overline{R}_2(-\zeta) \cup L^2$ where L^2 is an articulated ray stemming from B_1 , and $N(L^2) = (\overline{R}_1(\zeta) \cup \overline{R}_2(-\zeta)) \cup L$ by the periodicity condition. Set $L^0 = L$, $L^1 = N(L)$ and $L^i = \bigcup_{k \ge 0} \overline{l}^i_k$ for $i \in \{0, 1, 2\}$ (L^2 is as defined before). The relations $N(l_0^1) = N(l_1^2) = R_2(-\zeta)$ and also $N(l_1^1) = N(l_0^2) = R_1(\zeta)$ determine the first rays of L^1 and L^2 : $l_0^2 = R_1(\zeta/2), \ l_0^1 = R_2(-\zeta/2), \ l_1^2 = R_2'(-\zeta), \ l_1^1 =$ $R'_1(\zeta)$. The similar identities $N(l^0_k) = l^1_k = N(l^1_{k+2})$ and $N(l^2_{k+2}) = l^0_k$ will guarantee uniqueness provided we have inverse branches for N and know which one to choose. This will be shown in the next step by proving that $L^0 \cup L^1$ and L^2 are contained in $U_2 \cup N^{-1}(\{b_1, b_2\})$ and $U_1 \cup N^{-1}(\{b_1, b_2\})$ respectively, and by observing that N is univalent on those sets.

2. Localization. First, the articulated rays L^i all lie in $V \cup \{b_1, b_2\}$. Indeed, by step 1, each articulated ray starts with \overline{l}_0^i in $V \cup \{b_1, b_2\}$. Moreover, if L^i crosses ∂V , there exists j so that $2^j \theta = \zeta$ or $2^j \zeta = 0$ (recall that L^i consists only of preimages of $R_1(\zeta)$ and $R_2(-\zeta)$, and that two rays in B_k cannot land at the same point). This is impossible since θ is dyadic but not ζ . The following relations $N(L^0) = L^1, N(L^1) = L^2 \cup \overline{R}_1(\zeta) \cup \overline{R}_2(-\zeta)$ and $N(L^2) = L \cup \overline{R}_1(\zeta) \cup \overline{R}_2(-\zeta)$ imply that $N(L^i)$ all lie in $V \cup \{b_1, b_2\}$. Therefore L^0 and L^1 lie in $U_2 \cup N^{-1}(\{b_1, b_2\})$ (just because they start there by step 1) while L^2 lies in $U_1 \cup N^{-1}(\{b_1, b_2\})$.

3. Existence. Let $g_i: V \to U_i$ denote the inverse of the homeomorphism $U_i \to V$ induced by N. The articulated rays L_i are constructed as follow. Their rays of depth 0 and 1 are determined in point 1. The other rays are obtained inductively by applying the relevant inverse branch to each additional relation of step 1: $l_{k+2}^2 = g_1(l_k^1), \ l_k^0 = g_2(l_k^1), \ l_{k+2}^0 = g_2(l_k^2)$. The curves L^0, L^1 and L^2 constructed this way are clearly articulated rays.

We will now prove that L^0 converges to y. For this, we rewrite the above relations as $l_{k+2}^i = h_i(l_k^i)$ where $h_0 = g_2g_1g_2$, $h_1 = g_2^2g_1$ and $h_2 = g_1g_2^2$ are as defined on V. We claim that $\overline{h_i(V)} \subset V$. Indeed, $h_i(V)$ is included in $g_1(U_2)$ or $g_2(U_1)$. Moreover, $\partial g_1(U_2) \cap \partial V \subset \overline{R}_1(0) \cup \overline{R}_3(0)$ because $g_1(U_2) \subset U_1$. But $\overline{R}_1(0)$ is disjoint from ∂U_2 and hence also from $\partial g_1(U_2)$. On the other hand, $\overline{R}_3(0)$ cannot meet $\partial g_1(U_2)$; otherwise the latter would not be connected (the only rays converging to ∞ are the $R_i(0)$). Hence, $\overline{g_1(U_2)} \subset V$ and similarly $\overline{g_2(U_1)} \subset V$.

The map $h_0: V \to V$ contracts the hyperbolic metric by a bounded factor, so that the hyperbolic lengths of the rays l_k^0 decrease geometrically and L^0 converges. Moreover, the limit point is a fixed point of h_0 in U_2 . Now, h_0 has at most one fixed point and, considering the action of h_0 on the rays of $B_3 \cap V$, we see that $h_0(y) = y$. Therefore, L converges to y.

COROLLARY 4.4. Let L be the articulated ray of Proposition 4.3. Then L, N(L) and $N^2(L)$ do not intersect outside $\{b_2\}$.

Proof. In the previous proof (Proposition 4.3 point 2.) it is shown that L^0 and L^1 lie in $U_2 \cup N^{-1}(\{b_1, b_2\})$ while L^2 lies in $U_1 \cup N^{-1}(\{b_1, b_2\})$. Hence L^1 and L^2 , as well as L^0 and L^2 , do not intersect. If L^0 and L^1 intersect, their images L^1 and $\overline{R}_1(\zeta) \cup \overline{R}_2(-\zeta) \cup L^2$ also do. However, this is not possible out of b_2 . Hence L^0 and L^1 can intersect only at depth 1, i.e. at $N^{-1}(\{b_2\})$. But l_1^0, l_2^0 are not in B'_2 , so that L^0 and L^1 only intersect at b_2 .

Remark 4.5. The orbit of the articulated ray L avoids the critical point x_0 .

This is just by construction: L, L^1, L^2 belong to $U_1 \cup U_2$ and the orbit of $\overline{R}_1(\zeta) \cup \overline{R}_2(-\zeta)$ avoids x_0 .

5. Graphs defining puzzles

In this section we define graphs in $\overline{X} = \widehat{\mathbf{C}} \setminus \bigcup_{i=1,2,3} \phi_i(e^{-v}\overline{\mathbf{D}})$ (for any v > 0) that turn out to be useful puzzles for well chosen angles. At the end

of the section we establish that the closure of the puzzle pieces intersect ∂B_i under a connected set.

5.1. Puzzles.

Definition 5.1. Let $\zeta \in G \cap [\alpha, 2\alpha)$ be a nondyadic rational angle and L the articulated ray constructed in Proposition 4.3 stemming from B_2 with angle $-\zeta/4$. Given θ, η in $G \cap \mathbf{Q}/\mathbf{Z}$, we define two graphs $I(\theta)$ and $II(\zeta, \eta)$ as follows:

$$I(\theta) = \partial X \cup \left(\overline{X} \cap \left(\overline{R}_1(0) \cup \overline{R}_2(0) \cup \overline{R}_3(0) \cup \bigcup_{j \ge 0} \left(\overline{R}_1(2^j\theta) \cup \overline{R}_2(-2^j\theta) \right) \right) \right);$$
$$II(\zeta, \eta) = \partial X \cup \left(\overline{X} \cap \left(\bigcup_{j \ge 0} \left(N^j(L) \cup \overline{R}_1(2^j\eta) \cup \overline{R}_2(-2^j\eta) \cup \overline{R}_3(2^j/7) \right) \right) \right).$$



Figure 7: On the left the graph $I(\theta)$, on the right $II(\zeta, \eta)$.

Remark 5.2. The cycle of cut rays generated by $R_1(\eta) \cup R_2(-\eta)$ is added in order to (clearly) surround the critical point for the graph II (see Proposition 5.4).

Remark 5.3. The graphs considered in Definition 5.1 are connected and do not have terminal edges since $\theta, \eta \in G$ (see Figure 7).

For the purpose of the proof of Proposition 5.4 below, we recall some notation:

 $Q_i(t_1, t_2) = \overline{\{\phi_i(r \, e^{2i\pi t}), \ r \in [0, 1), t_1 \le t \le t_2\}} \text{ (see example 3.1);}$

for $2\kappa \in G$, with $\kappa \notin G$ (resp. $\kappa \in G$), $\gamma(\eta, \kappa)$ is the curve (resp. the two curves)

 $\overline{R}_1(\eta) \cup \overline{R}_2(-\eta) \cup \overline{R}_1(\kappa) \cup \overline{R}_2(-\kappa) \cup \overline{R}_1'(2\eta) \cup \overline{R}_2'(-2\eta) \cup \overline{R}_1'(2\kappa) \cup \overline{R}_2'(-2\kappa)$ (see Lemma 3.14 and Figure 4); and $\gamma(\theta) = \overline{R}_1(\theta) \cup \overline{R}_2(-\theta) \cup \overline{R}_1(0) \cup \overline{R}_2(0)$ for $\theta \in G$.

PROPOSITION 5.4. Let $\theta = 1 - \frac{1}{2^n - 1}$, $\eta = 1 - \frac{1}{2^m - 1}$ and $\zeta = \frac{1}{2^i}(1 - \frac{1}{2^r - 1})$. For n, m, r large enough $\theta, \eta, 2^i \zeta$ are in G; and for i properly chosen, the graphs $I(\theta)$ and $II(\zeta, \eta)$ are puzzles surrounding the critical point x_0 . Moreover, for n > m > r large enough, every point of $\partial B_1 \cup \partial B_2$ is surrounded by $I(\theta)$ or $II(\zeta, \eta)$ at a uniformly bounded depth.

Proof. We focus here on ∂B_1 (the argument is similar for ∂B_2).

Step 1. For n, m, r large enough $\theta, \eta, 2^i \zeta$ are in G, and for i properly chosen, $I(\theta)$ and $II(\zeta, \eta)$ are puzzles. Indeed, for n, m, r large enough, θ, η and $1 - 1/(2^r - 1)$ belong to G by Corollary 3.16 and, for some i (depending on r) $\zeta = 1/2^i(1 - 1/(2^r - 1))$ lies in $(\alpha, 2\alpha)$. By definition the graphs satisfy the condition $\Gamma \cap X' \subset f^{-1}(\Gamma)$. Moreover, if the orbit of the critical point meets one of the graphs, it must necessarily be at the landing point of a ray. The critical point would then follows this periodic ray but this is avoided by changing n, m, r to larger values.

Step 2. Every point of $X' \setminus N^{-1}(I(\theta))$ sitting in $Q_1(\theta/4, \theta + 1/2)$ is surrounded by $I(\theta)$ at depth 0. Let x be such a point. The piece $P_0(x)$ is the connected component of $X \setminus \gamma(\theta/2)$ intersecting $R_1(1/4)$ and $P_1(x)$ is included in a bounded connected component of $X' \setminus \gamma(\theta+1/2, \theta/4)$ since $x \in Q_1(\theta/4, \theta+1/2)$ (see Figure 8 below).

Hence $\overline{P}_1(x) \subset P_0(x)$ since any point of $\gamma(\theta + 1/2, \theta/4)$ which is also on $\partial P_0(x)$ would be critical (Lemma 3.8 and trivial fact 2.6) but there is no periodic critical point on J(N).

Step 3. For n large enough the critical point x_0 is surrounded by $I(\theta)$. By Remark 2.16 and Step 2 above, it suffices to show that $N^k(x_0)$ is for some k in D the bounded connected component of $\mathbb{C} \setminus \gamma(\theta + 1/2, \theta/4)$. Let k be the smallest integer such that $2^k \alpha \in [1/4, 1/2)$. For n large, $2^k \alpha < \theta + 1/2 < \theta/2 < 1/2$. Hence, $\alpha < (\theta + 1/2)/2^k < \theta/2^{k+1} \le 2\alpha$ and Lemma 3.14 insures that x_0 belongs to the bounded component D_k of $\widehat{\mathbb{C}} \setminus \gamma(\frac{\theta + 1/2}{2^k}, \frac{\theta}{2^{k+2}})$. Now, looking at the image of the rays in B_1 , one sees that N^k maps D_k inside D.

Step 4. Every point of $X' \setminus N^{-1}(II(\zeta, \eta))$ sitting in $Q_1(\epsilon/2 + 1/2, \zeta/4)$ is surrounded by $II(\zeta, \eta)$ at depth 0, where $\epsilon = \sup_{i\geq 0} \{2^i \zeta \mod 1, 2^i \eta \mod 1\}$. Let x be such a point. The piece $P_0(x)$ is the unbounded connected component of



Figure 8: The graph $N^{-1}(I(\theta))$ dashed.

 $\widehat{\mathbf{C}} \setminus II(\zeta, \eta)$. It is very easy to see that $\partial P_1(x) \cap \partial P_0(x)$ is empty. Indeed, $P_1(x)$ is the unbounded connected component of $N^{-1}(P_0(x))$. Hence its boundary consists of the rays $R_1(1/2 + \epsilon/2)$, $R_2(1/2 - \epsilon/2)$, $R_3(1/14)$, $R_3(11/14)$ and the preimage of L stemming from B_2 with angle $-\zeta/8$ together with the preimage of L^2 stemming from B_1 with angle $\zeta/4$ (L^2 is the articulated ray in N(L) and stems from B_1 ; see Figure 5).

Step 5. The critical point x_0 is surrounded by $II(\zeta, \eta)$ provided m > r. Let $\zeta_1 = 1 - \frac{1}{2^r - 1}$. If m > r then $\zeta_1 < \eta < 1$, so that $\zeta_1/4 < \eta/4 < \zeta_1+1/2 < \zeta_1/2$. Hence the piece $P_1(x_0)$ is included in D' the bounded connected component of $X' \setminus \gamma(\eta/4, \zeta + 1/2)$. It is easy to see that $\overline{D'} \cap II(\zeta, \eta)$ is empty. The end of the argument goes as in Step 3: if k denotes the smallest integer such that $2^k \alpha \in [1/4, 1/2)$, for m, r large enough $N^k(x_0) \in D'$.

Step 6. For n, m, r, i such that Steps 1–5 hold, every point of ∂B_1 is surrounded by $I(\theta)$ or $II(\zeta, \eta)$ at a uniformly bounded depth. The points of $\partial B_1 \cap I(\theta)$ are mapped to the landing point z of $R_1(\theta)$. For n large, $\epsilon/2+1/2 < \theta$, so z belongs to $Q_1(\epsilon/2+1/2, \zeta/4)$ and by Step 4 and Remark 2.16 the points of $\partial B_1 \cap I(\theta)$ are surrounded by $II(\zeta, \eta)$ at uniformly bounded depth. \Box

To conclude, we now show that ∂B_1 is covered, for some p, by the subsets $N^{-i}(\Delta), 0 \leq i \leq p$, where $\Delta = Q_1(\theta/4, \theta+1/2) \cup Q_1(\epsilon/2+1/2, \zeta/4)$. First, for n large enough, $\epsilon/4 + 1/4 < \theta + 1/2 < 1/2$, so $\Delta \cup N^{-1}(\Delta)$ covers $Q_1(\theta/4, 1/2)$. Let p_0 be such that $0 < \theta/(4.2^{p_0}) < \zeta/4$; then the sets $Q_1(\epsilon/2 + 1/2, \zeta/4)$ and $N^{-i}(Q_1(\theta/4, 1/2)), 0 \leq i \leq p_0$, cover $Q_1(\epsilon/2 + 1/2, 1/2)$. Finally for

P. ROESCH



Figure 9: The graph $N^{-1}(II(\zeta, \eta))$ dashed, with the piece $P_1(x)$ of step 4. p_1 satisfying $\epsilon/2 + 1/2 < 1 - 1/2^{p_1} < 1$, the sets $N^{-i}(Q_1(\epsilon/2 + 1/2, 1/2)),$ $0 \le i \le p_1$, cover ∂B_1 entirely.

5.2. Connectivity of the neighborhoods. We establish now the connectivity of $\overline{P}_n(x) \cap \partial B_i$, where P_n is any depth n puzzle piece for $I(\zeta)$ or $II(\zeta, \eta)$.

We prove inductively that for x in ∂B_i , any puzzle piece $\overline{P}_n(x)$ cuts ∂B_i along some $\partial B_i \cap Q_i(t_1, t_2)$. This intersection is connected, as seen in Example 3.1 (where $Q_i(t_1, t_2)$ is defined).

LEMMA 5.5. Let P be an open disc, R any connected component of $N^{-1}(P)$ intersecting B_i , $i \in \{1, 2, 3\}$. Under condition (*) or (**), $\overline{P} \cap \overline{B}_i = Q_i(\theta_1, \theta_2)$ implies that also $\overline{R} \cap \overline{B}_i = Q_i(\eta_1, \eta_2)$ for $\{\eta_1, \eta_2\} = \{\theta_1/2, \theta_2/2\}$ or $\{\theta_1/2 + 1/2, \theta_2/2 + 1/2\}$.

Condition (*) respectively (**) is that no critical point belongs to R, respectively that $R \cap B_i \subset Q_i(t, t')$ with $|t - t'| \leq 1/2$.

Proof. By assumption $\overline{P} \cap \overline{B}_i = Q_i(\theta_1, \theta_2)$, and so the set $R \cap B_i \subset N^{-1}(P) \cap B_i$ lies in $N^{-1}(Q_i(\theta_1, \theta_2)) \cap \overline{B}_i$. But $N^{-1}(Q_i(\theta_1, \theta_2))$ intersects \overline{B}_i under the set $\Delta_1 = Q_i(\theta_1/2, \theta_2/2)$ and $\Delta_2 = Q_i(\theta_1/2 + 1/2, \theta_2/2 + 1/2)$ since N is conjugated on B_i to $z \mapsto z^2$. So $R \cap B_i \subset \Delta_1 \cup \Delta_2$ and R intersects Δ_1 or Δ_2 . If R intersects Δ_j it contains $\operatorname{Int}\Delta_j$ since N induces a covering $R \to P$ and $\operatorname{Int}\Delta_j \subset N^{-1}(P)$. But R cannot contain Δ_1 and Δ_2 because of condition (*) or (**). Indeed, under condition (**) Δ_1 and Δ_2 are opposite and under condition (*) $N: R \to P$ would be (at least) a double cover over a disc and so ramified in R. Hence $R \cap B_i = \operatorname{Int}\Delta_1$ or $\operatorname{Int}\Delta_2$ so that $\overline{R} \cap \partial B_i = \Delta_1 \cap \partial B_i$ or $\Delta_2 \cap \partial B_i$. COROLLARY 5.6. For any depth n piece $P_n(x)$ (of the puzzles $I(\theta)$ or $II(\eta, \zeta)$ and $x \in \partial B_i$), $\overline{P}_n \cap \partial B_i$ is connected; more precisely $\overline{P}_n \cap \overline{B}_i = Q_i(\theta_n, \theta'_n) \cap \overline{X}_n$ for i = 1, 2 and some θ_n, θ'_n .

Proof. Every piece $P_n(x)$ satisfies the assumption of Lemma 5.5 under condition (**). Indeed, the pieces $P_n(x)$ are discs and intersect B_i ; they are all included in $P_0(x)$ which satisfies condition (**) by choice of θ, η and ζ .

6. Renormalizations

Definition 6.1. Let $f: \widehat{\mathbf{C}} \to \widehat{\mathbf{C}}$ be a rational map and x_0 a critical point of f. Let k > 1. The map f is k-renormalizable at x_0 , if f^k induces, between some discs Y' and Y containing x_0 , a quadratic-like map (i.e. a polynomiallike map of degree two) whose filled Julia set $K(f^k)$ is connected (see Definition 2.11). The map $f^k: Y' \to Y$ is called a k-renormalization of f in x_0 . The smallest integer k for which f is k-renormalizable at x_0 is called the minimal renormalization level.

Notation 6.2. If $f^k: Y' \to Y$ is a k-renormalization of f in x_0 , the straightening Theorem (see [D-H2]) shows that f^k is conjugated, by a quasi-conformal map σ , to a unique quadratic polynomial $f_c(z) = z^2 + c$ ($c \in \mathbf{C}$) on a neighborhood of $K = K(f^k)$. The filled Julia set of f_c will be denoted by $K(f_c)$, and $K = \sigma^{-1}(K(f_c))(=K_c$ if there is some ambiguity). If $c \neq 1/4$, the polynomial f_c has exactly one repelling fixed point $\beta(f_c)$ that does not disconnect $K(f_c)$ (see [McM1]), in the terminology of [D-H1] it is the point of external argument 0. Let $\alpha(f_c)$ be the other fixed point $(\beta(f_{1/4}) = \alpha(f_{1/4}))$ and $\beta'(f_c) \neq \beta(f_c)$ the other preimage of $\beta(f_c)$. Similarly we note $\beta = \beta_c = \sigma^{-1}(\beta(f_c))$.

6.1. Linearizability question.

THEOREM 5. If N possesses a periodic point x of multiplier $\lambda = e^{2i\pi\alpha}$ with $\alpha \in \mathbf{R}$ then

- (1) N is renormalizable;
- (2) $\alpha \in \mathcal{B}$ if and only if N is linearizable near x;
- (3) if $\alpha \notin \mathcal{B} \cup \mathbf{Q}$ there exist periodic cycles in any neighborhood of x.

Proof. Let k be the period of x. We first prove that N is k-renormalizable and then that the renormalized polynomial has the desired properties.

For any puzzle Γ of Definition 5.1, the puzzle pieces $P_n(x)$ are well defined since x is not repelling (and so $x \notin \Gamma_n = N^{-n}(\Gamma)$ for all $n \ge 0$). There is some point x' in the orbit of x such that the puzzle pieces $P_n(x')$ contain a critical point x_0 , i.e. $P_n(x') = P_n(x_0)$ for $n \ge 0$. Indeed, if for all x' in the orbit of x, the piece $P_n(x')$ is not critical, the restriction $N^k \colon P_{n+k}(x) \to P_n(x)$ would be invertible, and by Schwarz' lemma, x would be an attracting fixed point of the inverse (since for some n, $P_{n+1}(x) \neq P_n(x)$). Let p be such that $\overline{P}_{p+1}(x_0) \subset P_p(x_0)$ (Proposition 5.4 provides graphs satisfying this); the restriction $N^k \colon P_{p+k}(x) \to P_p(x)$ is then a quadratic like map.

Let σ be the straightening map and $f_c(z) = z^2 + c$ the quadratic polynomial conjugated to N^k by σ (see Notation 6.2). The point $\sigma(x)$ is fixed by f_c of multiplier λ since the multiplier of an indifferent fixed point is a quasiconformal invariant (see [N]). Finally by Yoccoz' result (see [Y]) we obtain that $\alpha \in \mathcal{B}$ if and only if N is linearizable near x and, if $\alpha \notin \mathcal{B}$ there exist small cycles near x.

6.2. Renormalization via puzzle pieces. More generally we prove here (Lemma 6.5) that if N is renormalizable we can take puzzles pieces as sets of renormalization.

LEMMA 6.3. Let N be k-renormalizable at x_0 with filled Julia set $K = K(N^k)$. If K intersects \overline{B}_1 (resp. \overline{B}_2) then there exists exactly one ray in B_1 (resp. B_2) accumulating on K. It has angle $p/(2^k - 1)$ with $p = p(N) < (2^k - 1)/2$ (resp. angle $1 - q(N)/(2^k - 1) > 1/2$) and converges to the fixed point β_c of $K = K_c$.

Proof. We consider the situation where $K \cap \overline{B}_1 \neq \emptyset$. Let V_0 be the connected component of $\widehat{\mathbb{C}} \setminus (\overline{R}_1(0) \cup \overline{R}_1(1/2) \cup \overline{R}_2(0) \cup \overline{R}_2(1/2))$ containing x_0 . Then K is included in V_0 since it contains x_0 and cannot cross the eventually fixed ∂V_0 . Since $N^k(K) = K$, one can consider for every $n \ge 0$ the connected component V_n of $N^{-k}(V_{n-1})$ containing K. They form a decreasing sequence of open sets: $N^k(\partial V_0) \cap V_0 = \emptyset$; hence $\partial V_0 \cap V_1 = \emptyset$ so that V_1 is included in V_0 and then $V_{n+1} \subset V_n$. After Lemma 5.5 under the condition (**), $\overline{V}_n \cap \overline{B}_1 = Q_1(\theta_n, \theta'_n)$ for some $\theta_n < \theta'_n$ such that $|\theta'_{n+1} - \theta_{n+1}| = 1/2^k |\theta'_n - \theta_n|$. The common limit $\kappa = \lim \theta_n = \lim \theta'_n$ is of the type $p/(2^k - 1)$. Indeed, since $N^k(V_{n+1}) = V_n$ for every n, necessarily $\theta_n \le \{2^k \kappa\} \le \theta'_n$ (where $\{t\}$ is the fractional part of t), so that $\{2^k \kappa\} = \kappa$. Hence, the ray of angle κ lands at a point which is fixed by N^k . Necessarily this point is β_c for $K = K_c$ since $\beta(f_c)$ is the unique fixed point of f_c where there is an external fixed access (Theorem A of [Pe2]).

The unicity of κ follows from the remark that any ray $R_1(\theta)$ accumulating K_c should belong to V_n so that θ is between θ_n and θ'_n for every n. Hence $\theta = \kappa$.

COROLLARY 6.4. If N is k-renormalizable at x_0 and if $K = K(N^k)$ intersects \overline{B}_1 and also \overline{B}_2 then the unique rays accumulating on K are $R_1(\alpha)$ and $R_2(-\alpha)$ where α is the Head angle.

Proof. If $R_1(\theta_1)$ and $R_2(\theta_2)$ accumulate on $K = K_c$, they both converge to β_c (Lemma 6.3) so that $\theta_1 = 1 - \theta_2 \in G$. If there exists $\theta < \theta_1$ in G, K lies in the bounded component of $\widehat{\mathbf{C}} \setminus (\overline{R}_1(1/2) \cup \overline{R}_2(1/2) \cup \overline{R}_1(\theta) \cup \overline{R}_2(-\theta))$. Indeed, it contains $R_1(\theta_1)$ and the curve cannot accumulate on K after Lemma 6.3. This contradicts Lemma 3.14 which asserts that the critical point x_0 is in the unbounded component (see Figure 4).

LEMMA 6.5. If N is k-renormalizable at x_0 then for any puzzle Γ of the form $I(\theta)$ or $II(\eta, \zeta)$ satisfying Proposition 5.4 the following holds (where $P_n(x_0)$ denote the critical puzzle pieces):

- (1) For l sufficiently large $N^k(P_{l+k}(x_0)) = P_l(x_0)$ and $\operatorname{Imp}(x_0) \supset K(N^k)$;
- (2) There exists j dividing k such that, for l large, $N^j \colon P_{l+j}(x_0) \to P_l(x_0)$ is a renormalization of N (in x_0) and $\text{Imp}(x_0) = K(N^j)$. The level j is the minimal renormalization level.

Proof. 1. The filled Julia set $K(N^k)$ is included in $\overline{P}_n(x_0)$. Else the boundary $\partial P_n(x_0)$ would disconnect $K(N^k)$. Then after iterations, two rays of Γ in $B_1 \cup B_2$ would land at the same point of $K(N^k)$ ($\partial B_3 \cap K(N^k)$) $= \emptyset$ after Lemma 3.14) and their closure would also disconnect $K(N^k)$ (after Corollary 3.9 and since the critical orbit is disjoint from Γ). But Lemma 6.3 insures that those rays are fixed by N^k and have to land at β_c . Taking the image by the local homeomorphism σ would imply that $\beta(f_c)$ disconnects the quadratic Julia set $K(f_c)$ which is impossible (see [McM1, Thm 6.10]). Hence $K(N^k) \subset \overline{P}_n(x_0)$ and therefore $K(N^k) \subset \text{Imp}(x_0)$. Moreover $N^k(P_{l+k}(x_0))$ is a piece containing $N^k(K) = K$, of depth l; so it is $P_l(x_0)$.

2. Proposition 5.4 gives an l_0 such that $\overline{P}_{l_0+1}(x_0) \subset P_{l_0}(x_0)$. Let j be the first integer in [0,k] such that $N^j(P_{l_0+j}(x_0)) = P_{l_0}(x_0)$, then the map $N^j: P_{l_0+j}(x_0) \to P_{l_0}(x_0)$ is a renormalization of N (in x_0). Taking the inverse images by $N, \overline{P}_{l+j}(x_0) \subset P_l(x_0)$ for all $l \ge l_0$ (see Remark 2.16) so that $N^j: P_{l+j}(x_0) \to P_l(x_0)$ is also a renormalization of N for l large enough. Finally $\operatorname{Imp}(x_0) = K(N^j)$ by definition of $K(N^j)$.

Assume that j is not the minimal renormalization level. Then, there exists r < j such that $N^r(K(N^r)) = K(N^r)$. From 1, $K(N^r)$ is included in $\operatorname{Imp}(x_0) \subset \overline{P}_l(x_0)$. But this contradicts the minimality in the definition of j since $N^r(P_{l+r}(x_0))$ contains x_0 .

If j does not divide k, there exists m such that k = ij + m with 0 < m < j. Then $N^m \colon P_{l+m}(x_0) \to P_l(x_0)$ would be a renormalization of N in x_0 . This contradicts the minimality of j.

COROLLARY 6.6. If N is k-renormalizable at x_0 and if k is the minimal level of renormalization, then the images $N^i(K(N^k))$ are disjoint for $0 \le i \le k-1$.

P. ROESCH

Proof. Let Γ be a puzzle of the form $I(\theta), II(\eta, \zeta)$ satisfying Proposition 5.4. For some large $l, \overline{P}_l(x_0) \subset P_{l-1}(x_0)$. Let $i \leq k-1$ be the first integer such that $N^i(K(N^k))$ intersects $K(N^k)$; then $N^i(P_l(x_0)) = P_{l-i}(x_0)$ and the restriction $N^i \colon P_l(x_0) \to P_{l-i}(x_0)$ is an *i*-renormalization of N. This contradicts the minimality of k.

7. The boundaries ∂B_i are Jordan curves

In this section we prove that ∂B_i is locally connected. We will use the previous sections. The case of B_3 is really simpler and treated separately (Section 7.1). We distinguish when N is renormalizable around x_0 or not.

As pointed out in Remark 2.4 (using Remarks 2.2 and 3.10 and Lemma 3.8), the local connectedness of ∂B_i has for consequence that $\partial B_i = \phi_i(\mathbf{S}^1)$ (where ϕ_i denotes also the extension) is a Jordan curve formed by the landing points of the rays, whenever it is not conjugated to the polynomial $z^3 + 3i/\sqrt{2}z^2$. Hence in this section we prove that ∂B_i is locally connected for a genuine Newton map.

7.1. The boundary ∂B_3 is a Jordan curve. The orbit of the critical point x_0 is "far from" \overline{B}_3 (Lemma 3.14), so it is rather easy to prove by hand (without Yoccoz' Theorem) the following:

PROPOSITION 7.1. The boundary of B_3 is a Jordan curve.

Proof. There exists j such that $t = \frac{1}{2^j} \in [\alpha, 2\alpha)$. Denote by C the curve $R_3(1/2) \cup \bigcup_{i=1,2} \overline{R}_i(\pm t/2) \cup \overline{R}'_i((-1)^{i+1}t) \cup \overline{R}'_i(0)$ and recall that

$$\overline{X} = \widehat{\mathbf{C}} \setminus \bigcup_{i=1,2,3} \phi_i(e^{-v}\overline{\mathbf{D}}) \quad \text{(for any } v > 0\text{)}.$$

1) a) Local connectivity of ∂B_3 at ∞ : The unbounded connected component P_0 of $X \setminus C$ is a disc (by choice of t). It is easy to check that P_1 , the connected component of $N^{-1}(P_0)$ containing ∞ , satisfies $\overline{P}_1 \subset P_0$ and that N induces a homeomorphism $P_1 \to P_0$ (Lemma 3.14); let $g: P_0 \to P_1$ be its inverse. Since $P_n = g^n(P_0)$ reduces to ∞ (Schwarz Lemma), so does $\overline{P}_n \cap \partial B_3$. Since $P_n \subset P_1$, no critical point belongs to P_n (by Lemma 3.14) so the sequence $\overline{P}_n \cap \partial B_3$ forms a basis of connected neighborhoods of ∞ and condition (*) of Lemma 5.5 is satisfied (the proof is the same as in corollary 5.6 even if P_n is not a puzzle piece).

b) Local connectivity of ∂B_3 at the preimages of ∞ : For $x \in N^{-k}(\infty)$, let R_n be the connected component of $N^{-k}(P_n)$ containing x, where P_n is as defined in a). For large enough n, the piece P_n satisfies the condition (**) of Lemma 5.5; therefore the sequence $\overline{R}_n \cap \partial B_3$ is a basis of connected neighborhoods of x (if $n \ge k+1$).



Figure 10: On the left illustration of 1a), on the right of case 2

2) Local connectivity of ∂B_3 outside the preimages of ∞ : Now we denote by P_0 the connected component of $X \setminus N(C)$ intersecting B_3 . It is easy to check that any connected component P_n of $N^{-n}(P_0)$ intersecting B_3 lies in $N^{-1}(P_0)$ by the fact that $\partial N^{-1}(P_0) \cap P_n = \emptyset$ (or equivalently that $N^{n-1}(C) \cap P_0 = \emptyset$) and that $N^{-1}(P_0)$ intersects P_n since $N^{-1}(P_0) \supset B_3 \cap X_1$. Therefore N^n induces a homeomorphism $P_n \to P_0$ since there is no critical point in $N^{-1}(P_0)$ (Lemma 3.14).

Let P_1 be the connected component of $N^{-1}(P_0)$ intersecting $R_3(3/4)$. The connected component P_2 of $N^{-1}(P_1)$ intersecting $R_3(3/8)$ is then compactly contained in P_0 since $\partial P_2 \cap (B_1 \cup B_2) = \emptyset$. Therefore, if we denote by $P_n(x)$ (resp. $P_{n+2}(x)$) the connected component of $N^{-n}(P_0)$ (resp. $N^{-n}(P_2)$) containing x, $A_n(x) = P_n(x) \setminus \overline{P}_{n+2}(x)$ is conformally equivalent to $P_0 \setminus \overline{P}_2$. Assuming that there exists a sequence n_j diverging to ∞ such that $N^{n_j}(x) \in Q = \overline{B}_3 \cap P_2$, we see that the annuli $A_{n_j}(x)$ form a sequence surrounding x and with the same modulus (as $P_0 \setminus \overline{P}_2$); one extracts a subsequence to obtain disjoint annuli and by Grötszch inequality $\operatorname{mod}(P_0 \setminus \overline{P}_{n_{k_j}}) \to \infty$, so that $\cap P_n$ reduces to a point.

Now we find the sequence n_j . For $x \in \partial B_3$ and $n \ge 0$, x belongs to some $Q_3(\theta_n, \theta'_n)$ with θ_n, θ'_n of the form $k_n/2^n, (k_n+1)/2^n$, since x is not a preimage of ∞ . Hence the common limit $\theta = \lim \theta_n = \lim \theta'_n$ satisfies $\theta_n < \theta < \theta'_n$ for every $n \ge 0$, and so is not dyadic. Moreover $N^n(x)$ and $N^n(R_3(\theta))$ belongs, or avoid, simultaneously, P_2 . Since a nondyadic angle θ has infinitely many occurrences

of 0, 1 in its dyadic expansion there exists n_j such that $N^{n_j}(x) \in Q_3(1/4, 1/2)$ and so n_j is in P_2 .

7.2. The boundary of B_1 and B_2 in the nonrenormalizable case.

PROPOSITION 7.2. If the cubic Newton map N is not renormalizable at x_0 , the boundaries ∂B_1 and ∂B_2 are Jordan curves.

Proof. Proposition 5.4 provides graphs $I(\theta)$ and $II(\eta, \zeta)$ satisfying the hypothesis of Yoccoz' Theorem for the set $C = \partial B_1 \cup \partial B_2$ (see Theorem 2.15). Since N is not renormalizable at x_0 , the second conclusion of the Theorem does not take place so that $\operatorname{Imp}(x_0) = \bigcap_{n\geq 0} \overline{P}_n(x_0) = \{x_0\}$ and $\operatorname{Imp}(y) = \bigcap_{n\geq 0} \overline{P}_n(y) = \{y\}$ for every $y \in \partial B_1 \cup \partial B_2$ (where $P_n(x)$ designs a piece of depth n for one of the previous graphs). Hence $\overline{P}_n(y) \cap \partial B_i$ form a basis of connected neighborhoods of y in ∂B_i by Corollary 5.6.

7.3. The boundary of B_1 and B_2 in the renormalizable case. We assume now that N is renormalizable at x_0 with minimal renormalization level k. We denote by $N^k: Y' \to Y$ a k-renormalization of N in x_0 , by $K = K(N^k)$ the connected filled Julia set of this renormalization and by σ the straightening map and by β the fixed point of N^k not disconnecting K (see Notation 6.2). By Lemma 6.5 we assume that Y and Y' are puzzle pieces.

Remark 7.3. If the boundary ∂B_1 , or ∂B_2 , does not intersect K, it is locally connected.

Proof. From Lemma 6.5 $\operatorname{Imp}(x_0) = K(N^k) = K$. Hence, the orbit of any point $x \in \partial B_i$ will never meet $\operatorname{Imp}(x_0)$ since it stays on ∂B_i . So by Theorem 2.15 $\operatorname{Imp}(x)$ reduces to x.

For this reason we assume in this section that $\overline{B}_1 \cap K \neq \emptyset$, the case of B_2 is analogous.

We define "sides" of K as follows. Let $R_1(\kappa)$ be the unique ray accumulating on K (Lemma 6.3). Its preimage $R'_1(2\kappa)$ converges to $\beta' \in K \cap N^{-1}(\beta)$ since the other preimage $R_1(\kappa + 1/2)$ cannot accumulate K (by unicity in B_1).

Definition 7.4. Let Δ_1 (resp. Δ_2) be the connected component of $\widehat{\mathbf{C}} \setminus \widetilde{K}$ containing B_2 (resp. B'_2), where

$$\tilde{K} = K \cup \overline{R}_1(0) \cup \overline{R}_3(0) \cup \overline{R}_3(1/2) \cup \overline{R}_1'(0) \cup \overline{R}_1'(2\kappa) \cup \overline{R}_1(\kappa).$$

LEMMA 7.5. The connected components Δ_1 and Δ_2 are discs.



Figure 11: On the left Δ_1, Δ_2 . On the right $\bigcup_i N^i(K \cup R_1(9/31))$ for N with Head angle 1/3 for the proof of Proposition 7.6.

Proof. This is a corollary of the following topological Lemma with the open disc $U = \widehat{\mathbf{C}} \setminus (\overline{R}_1(0) \cup \overline{R}_3(0) \cup \overline{R}_3(1/2))$ and the curves $a_1 = \overline{R}'_1(0) \cup \overline{R}'_1(2\kappa)$, $a_2 = \overline{R}_1(\kappa)$.

TOPOLOGICAL LEMMA. Let K be a connected, full compact set in a topological open disc U and a_1, \ldots, a_n simple disjoint closed arcs crossing K, as well as ∂U , only at one endpoint. Then the open set $U \setminus (K \cup a_1 \cup \cdots \cup a_n)$ is the union of exactly n topological discs.

Proof. Since $K \subset U$ is a connected, full compact set and U is a disc, there exists a homeomorphism $\varphi \colon U \setminus K \to \mathbb{C} \setminus \{0\}$. The compacts $b_i = \overline{\varphi(a_i \cap (U \setminus K))}$ are arcs of disjoint interior joining 0 to ∞ in $\widehat{\mathbb{C}}$. The open set $\widehat{\mathbb{C}} \setminus \bigcup_i b_i$ is then the union of n disjoint topological discs (Jordan's theorem) and is homeomorphic to $U \setminus (K \cup a_0 \cup \cdots \cup a_n)$.

The aim of this section is to prove:

PROPOSITION 7.6. For $i \in \{1, 2\}$ there exists an articulated ray $T_i \subset \Delta_i$ stemming from B'_i and converging to β .

COROLLARY 7.7. For any renormalization $N^j: Z' \to Z$ of N near x_0 , if $K(N^j)$ intersects ∂B_i , it is in exactly one point.

Proof. By Lemma 6.5, $K(N^j) \subset K$ so it is enough to prove that K intersects ∂B_i in one point. Recall that there is an angle κ such that the ray $R_1(\kappa)$ converges to the fixed point β of $N^k_{|K}$ (Lemma 6.3). The two articulated



Figure 12: Fixed articulated rays converging to β .

rays T_i , constructed in Proposition 7.6, also land at β . Hence $T = T_1 \cup T_2 \cup \overline{R}'_1(0) \cup \overline{R}'_2(0)$ is a Jordan curve separating $K \setminus \{\beta\}$ from $B_1 \cup B_2$. Indeed, T_i are stemming from B'_i , they are disjoint and stay in Δ_i . Finally, $(\overline{B}_1 \cup \overline{B}_2) \cap K$ is reduced to β .

COROLLARY 7.8. The boundaries ∂B_1 and ∂B_2 are Jordan curves.

Proof. Proposition 5.4 provides graphs $I(\theta)$ and $II(\eta, \zeta)$ satisfying the hypothesis of Yoccoz' Theorem (Theorem 2.15). Since N is renormalizable at x_0 by Lemma 6.5, we are in the second case of the conclusion of Yoccoz' Theorem for the graphs provided by Proposition 5.4; i.e. $N^k : P_{n+k}(x_0) \to$ $P_n(x_0)$ is a renormalization of N. By Corollary 7.7, $\partial B_i \cap K(N^k)$ is at most one point. Hence for any point $y \in \partial B_i$ whose orbit meets $K(N^k)$, the sequence $\overline{P}_n(y) \cap \partial B_i$ forms in ∂B_i a basis of connected neighborhoods of y. Indeed their intersection reduces to preimages of $K(N^k) \cap \partial B_i$, i.e. to points, and the connectedness follows from Corollary 5.6. On the other hand, if the orbit of $y \in \partial B_i$ does not meet $K(N^k) = \operatorname{Imp}(x_0)$, Yoccoz' Theorem implies that $\operatorname{Imp}(y) = \{y\}$. In both cases ∂B_i is locally connected at y.

Proof of Proposition 7.6. Let U be the connected component of $\widehat{\mathbf{C}} \setminus (\overline{R}_3(0) \cup \gamma(1/2))$ intersecting B_3 . It is a topological disc. Now applying k times the Topological Lemma to $N^j(K)$ and to the arcs $N^j(\overline{R}_1(\kappa))$ (for $0 \le j \le k-1$) one obtains that $V = U \setminus \bigcup_{0 \le j \le k} N^j(K \cup R_1(\kappa))$ is a topological disc.

The proof of the proposition is organized as follows.

- (1) Step 1 provides an inverse branch g_i of the renormalization $N^k : Y' \to Y$, which is defined on $\Delta_i \cap Y$, but extends to V and satisfies $g_i(V) \subset V \cap \Delta_i$.
- (2) In Step 2 we construct a finite articulated ray C_i , joining $b'_i = N^{-1}(b_i) \cap B'_i$ to $g_i(b'_i)$ and included in $\Delta_i \cap V$.
- (3) In Step 3 we define the articulated ray T_i as the union for $n \ge 0$ of $g_i^n(C_i)$; then $g_i(T_i) \subset T_i$ and T_i has finite length since g_i is a contraction in $Y \cap \Delta_i$.



Figure 13: Inverse branches

Step 1. There exists $g_i: V \to V \cap \Delta_i$ an inverse branch of N^k , such that the sequence $(g_i^n)_{n \in \mathbb{N}}$ converges for the open-compact topology of V to a constant $z_i \in K$.

To define an inverse branch of the renormalization N^k , we first define inverse branches of N on domains containing the orbit of K. There is no critical value of N in the disc $\Omega = \widehat{\mathbf{C}} \setminus \left(N(K \cup \overline{R}_1(\kappa)) \cup_{i=1,2,3} \overline{R}_i(0)\right)$ and since $N(N^j(K)) \subset \Omega$ for $1 \leq j \leq k-1$, let Ω_j be the connected component of $N^{-1}(\Omega)$ containing $N^j(K)$, where $N: \Omega_j \to \Omega$ admits an inverse branch called f_j . The case j = 0 requires more caution since N(K) is not in Ω . There are exactly three components forming $N^{-1}(\Omega)$, two are stuck to K, we denote by Ω_0^i the one contained in Δ_i (i = 1, 2) and by f_0^i the corresponding inverse branch. The maps f_i cannot be composed on Ω (since $f_i(\Omega) \not\subseteq \Omega$) but on $\widehat{\mathbf{C}} \setminus K'$ where $K' = \bigcup_{j \ge 0} N^j(K) \cup \overline{R}_1(0) \cup \overline{R}_2(0) \cup \overline{R}_3(0)$ (and in particular on $V \subset \widehat{\mathbf{C}} \setminus K'$) since $\widehat{\mathbf{C}} \setminus K' \subset \Omega$ and $f_i(\widehat{\mathbf{C}} \setminus K') \subset \widehat{\mathbf{C}} \setminus K'$, the second inclusion follows from the stability of K': $N(K') \subset K'$ and moreover $f_0^i(\widehat{\mathbf{C}} \setminus K') \subset \widehat{\mathbf{C}} \setminus K'$.

Now we can define g_i as the composition $g_i = f_0^i \circ f_1 \circ \cdots \circ f_{k-1} \colon V \to \Delta_i$; it is an inverse branch of N^k . By construction $g_i(V) \subset \widehat{\mathbf{C}} \setminus C'$ avoids $\gamma(1/2)$ since $f_0^i(\Omega) \subset N^{-1}(\Omega) \subset \widehat{\mathbf{C}} \setminus \gamma(1/2)$, so that $g_i(V) \subset V$ (since it is stuck to $K \subset V$).

Moreover, if $N^k \colon Y' \to Y$ is the k-renormalization in x_0 , with Y' small enough, then $g_i(Y \cap V) \subset Y'$. Indeed, $N^j(Y')$ is included in Ω for $2 \leq j \leq k$ and the image $f_j \circ \cdots \circ f_{k-1}(Y)$ is exactly $N^j(Y')$. Finally, since Y' is the only component of $N^{-1}(N(Y'))$ stuck to K, $f_0^i(N(Y') \cap \Omega)$ is included in Y' so that $g_i(V \cap Y') \subset Y' \subset Y$.

Since $g_i: V \to V$ is not surjective $(V \cap \Delta_i \neq V)$, the sequence of iterates $(g_i^n)_n$ converges uniformly on every compact set of V to a constant $z_i \in \partial V$ after the Denjoy-Wolff Lemma. The fact that $g_i(Y \cap V) \subset Y'$ forces the limit z_i to be in K.

Step 2. There exists a finite articulated ray $C_i \subset V \cap \Delta_i$ joining b'_i and $g_i(b'_i)$, for i = 1, 2.

The articulated ray C_1 is $C_1 = f_0^1(R_{k-1})$ where $(R_j)_{j \in \mathbb{N}}$ is a sequence of finite articulated rays stemming from B_1 or B_3 defined as follows (the construction is similar for C_2 with B_2 and B_3). Let $R_0 = R_3(1/2) \cup \overline{R}'_1(0)$ and, for $j \ge 0$, if $f_{k-1-j}(R_j)$ stems from $B_1 \cup B_3$, let R_{j+1} be $f_{k-1-j}(R_j)$. Otherwise take $R_{j+1} = f_{k-1-j}(R_j) \cup \overline{R}'_1(0) \cup R_3(1/2)$ or $R_{j+1} = f_{k-1-j}(R_j) \cup \overline{R}'_3(0) \cup R_1(1/2)$ depending on whether $f_{k-1-j}(R_j)$ stems from B'_1 or B'_3 . The sequence R_j is well defined since $R_j \subset \Omega$ and consists of finite articulated rays. Moreover, C_1 contains $g_1(b'_1)$ by construction and is located in $V \cap \Delta_1$ since it cannot cross the boundary of V nor Δ_1 (after Lemma 3.8 and Lemma 6.3).

To prove that C_1 stems from B'_1 , it is enough to show that R_{k-1} stems from B_1 with an angle $\theta < 2\kappa$ since C_1 is in Δ_1 . Let r < k be the largest integer such that $f_{k-1-(r-1)}(R_{r-1}) = R_r$ stems from B_3 . Then for r < j < k-1, $f_{k-1-j}(R_j)$ stems from B_1 , and precisely R_{r+1} stems from B_1 with $R_1(1/2)$ so that $2^{k-r-2}\theta = 1/2$ since $N^{k-r-2}(R_{k-1}) = R_{r+1}$. By definition of f_{k-1-r} , the curve $\gamma(1/2)$ does not separate $N^{k-1-r}(K)$ from $f_{k-1-r}(R_r)$. Therefore $1/2 < 2^{k-r-1}\kappa < 1$ and $\theta < 2\kappa$.

Step 3. The union $T_i = \bigcup_{n \ge 0} g_i^n(C_i)$ is an articulated ray located in $V \cap \Delta_i$. Moreover, T_i stems from b'_i , $N^k(T_i) \supset T_i$ and T_i converges to β . By construction $N^k(T_i) = T_i \cup N^k(C_i) \supset T_i$. After Step 1 and Step 2, $T_i \subset V \cap \Delta_i$ and T_i is an articulated ray stemming from b'_i ; it remains to show that T_i converges to β .

By Step 2, the sequence $(g_i^n)_n$ converges uniformly on any compact of V to a point $z_i \in K$. Since $\overline{C}_i \subset V$, the articulated ray T_i converges to z_i (since any accumulation point of T_i is z_i), and $N^k(z_i) = z_i$.

Let $y_i = \sigma(z_i)$ be the corresponding point in the model f_c , where σ (defined on Y) is the conjugacy between N^k and f_c . The arc $\sigma(T_i)$ (or at least a neighborhood of y_i) is a "fixed" access landing at $y_i \in K(f_c)$ (since $f_c(\sigma(T_i)) \supset$ $\sigma(T_i)$). But only $\beta(f_c)$ can be the landing point of a fixed external access to a periodic point of K, so that T_i lands at β .

7.4. Corollaries of the local connectivity of ∂B_i . Recall that we consider only genuine Newton maps.

LEMMA 7.9. If N has no Siegel point, its Fatou set is $\tilde{B}_1 \cup \tilde{B}_2 \cup \tilde{B}_3$ unless it is geometrically finite.

Proof. Sullivan's classification theorem gives us that, besides B_i , there can only be attracting or parabolic components in the Fatou set. Indeed, there is no Herman ring for J(N) is connected (see [Sh]). Moreover, the cycle defined by a component contains a critical point which is either x_0 or one of the roots b_1, b_2, b_3 . In the first case N is geometrically finite, in the second the components are in \tilde{B}_i .

THEOREM 6. For every cubic Newton map without Siegel disc, the boundary of the connected components of the Fatou set are Jordan curves.

Proof. If N is geometrically finite, the boundary is locally connected (see [D-H1, M2, T-Y]). Otherwise, the Fatou set is exactly the \tilde{B}_i (Lemma 7.9). Let U be any connected component of \tilde{B}_i . Its boundary is locally connected since ∂B_i is locally connected and N is a ramified covering.

We prove now that the boundary of a component U of one B_i is a Jordan curve. The boundary is a curve since the Böttcher coordinate extends (by Carathéodory's Theorem). This curve would not be Jordan if two rays land at the same point. For $U = B_j$ this is not possible after Remark 3.10 and Lemma 3.8. For any inverse image of B_j , iterating until B_j , one obtains that two rays in B_j land at the same point which is not possible.

If N is geometrically finite and U is not a component of B_i then N is renormalizable for any of the puzzles given by Proposition 5.4 and U is included in the sequence of critical puzzle pieces (see the proof of Theorem 5). Then $\sigma(U)$ is a Fatou component of a quadratic Julia set, where σ denotes the straightening map. So two rays in U cannot land at the same point (polynomial filled Julia sets are full). COROLLARY 7.10. There cannot be a Cremer point on the boundary of a component of the Fatou set F(N).

Proof. Assume that there is a Cremer point on the boundary of a component of F(N). Then N is not geometrically finite so F(N) is exactly the \tilde{B}_i (Lemma 7.9) or there is a Siegel point. Assume first that there is no Siegel point. Since the Cremer point x is a periodic point, it is on ∂B_i and it is accessible by a ray $R_i(\theta)$ (since ∂B_i is locally connected, Theorem 6 and Remark 2.4). Since two rays in B_i cannot land at the same point, $R_i(\theta)$ is periodic. But the landing point of a periodic ray is necessarily repelling or parabolic (Snail Lemma [M1]). If there is a Siegel point, then as in the proof of Theorem 5, N is renormalizable around x_0 and quasi-conformally conjugated to a quadratic polynomial having a Siegel disc. If there is a Cremer point on the boundary of the Siegel disc then by Naïhul's result there is a Cremer periodic point on the Siegel disc of the quadratic polynomial which is impossible (see [G-M]).

COROLLARY 7.11. N is topological conjugated on $\partial B_1 \cap \partial B_2$ to the multiplication by 2 on:

 $G = \{ \theta \in \mathbf{R} \mid \forall n \ge 0, 2^n \theta \mod 1 \in [\alpha, 1] \}.$

Moreover, the Head angle α belongs to G.

Proof. This is an immediate consequence from the fact that the Böttcher map ϕ_i extends to a homeomorphism between \overline{D} and \overline{B}_i .

8. Local connectivity of the whole Julia set

The aim of this section is to prove Theorem 1, Theorem 2 and Corollary 3, which provide surprising differences between rational maps and polynomials. First of all we prove Theorem 4. Then Proposition 8.3 gives a complete description of the situations where J(N) is locally connected. Finally, Sections 8.2 and 8.3 are devoted to the proof of the technical cases (1 and 2) of Proposition 8.3 that are not used for the previously mentioned results.

8.1. How rational maps differ from polynomials.

THEOREM 4. A cubic Newton map, without Siegel point, has a locally connected Julia set provided the orbit of the nonfixed critical point does not accumulate on the boundary of any fixed immediate basin of attraction.

To prove this Theorem we use the following characterization of locally connected sets, see [W, Th. 4.4, p. 113].

162

PROPOSITION 8.1. A connected compact set $J \subset \widehat{\mathbf{C}}$ is locally connected if and only if it satisfies the following properties:

- For every ε > 0, only a finite number of connected components of C \ J has a spherical diameter greater than ε;
- The boundary of every connected component of $\widehat{\mathbf{C}} \setminus J$ is locally connected.

We obtain the first condition by applying the so-called "shrinking Lemma". See [T-Y, Prop. A.3] or [L-M, Shrinking Lemma, p. 86]. This lemma is also an easy consequence of Mañe's theorem and Koebe's distortion theorem. Here the *post-critical set* P(f) is the closure of the forward orbit of the critical points of f.

PROPOSITION 8.2 (Shrinking Lemma). Let $f: \widehat{\mathbf{C}} \to \widehat{\mathbf{C}}$ be a rational map and D a topological disc whose closure \overline{D} does not cross the post-critical set of f. Then the following holds:

- Either \overline{D} is contained in a Siegel disc (i.e. Fatou component containing a Siegel point) or a Herman ring (i.e. Fatou component which is topologically a ring);
- Or, for every ε > 0, only a finite number of iterated preimages of D have a spherical diameter greater than ε.

Proof of Theorem 4. Unless N is geometrically finite, in which case J(N) is locally connected (see [D-H1], [M2], [T-Y]), the Fatou set is only the union of the \tilde{B}_i (Lemma 7.9). Hence the second condition of Proposition 8.1 is satisfied (Theorem 6). Since J(N) is connected (see [Sh]), we can apply Proposition 8.2 to $D = B'_i$ for $i \in \{1, 2, 3\}$ (since the orbit of the x_0 does not accumulate on the boundary of B_1, B_2, B_3 and there is no Siegel point). Hence J(N) fulfills the two conditions for being locally connected (Proposition 8.1).

PROPOSITION 8.3. Let N be a cubic Newton method without Siegel periodic points. The Julia set of N is locally connected in any of the following cases:

- 1) N is not renormalizable;
- 2) N is renormalizable (in x_0) exactly once and has no Cremer point;
- 3) N is renormalizable once and the filled Julia set of its renormalization does not encounter the closure of the immediate basins B_i ;
- 4) N is renormalizable at least twice.

P. ROESCH

Remark 8.4. The first case does not follow from Theorem 4 and will be proved later, as will the second case which is rather technical.

The dichotomy of this proposition is due to the following question 8.5.

Question 8.5. If N is exactly once renormalizable with a Cremer point, can the positive orbit of x_0 accumulate one of the \overline{B}_i ?

This is in fact equivalent to the following conjecture concerning quadratic polynomials:

CONJECTURE 8.6. If $f_c(z) = z^2 + c$ has a Cremer point, can the positive orbit of 0 accumulate the β fixed point (i.e. $\beta(f_c)$)?

Regarding the Newton method one can hope that the following is true:

CONJECTURE 8.7. Is the Julia set of a genuine cubic Newton method always locally connected?

Now we prove points 3 and 4 of Proposition 8.3.

LEMMA 8.8. Assume that N is renormalizable with minimal renormalization level k. Let f_c denotes the quadratic polynomial conjugated to N^k . The following equivalence holds: N is renormalizable exactly once if and only if f_c is not renormalizable.

Proof. Let $N^k \colon Y' \to Y$ be the renormalization of the minimal level of N and let σ denote the conjugacy between N^k and f_c .

Assume that f_c is renormalizable. Then there exists $i \in \mathbf{N}$ and open sets $U' \subset U \subset \sigma(Y')$ such that $f_c^i \colon U' \to U$ is quadratic-like. Since $N_{|_{Y'}}^{ik} = \sigma^{-1} \circ f_c^i \circ \sigma$, the restriction of $(N^k)^i$ on $\sigma^{-1}(U')$ defines a renormalization of N^k .

Conversely if N is *j*-renormalizable at x_0 , with $j \neq k$ then j = ik by Lemma 6.5. Hence N^k is *i*-renormalizable at x_0 with filled Julia set $K(N^j)$ and $K(N^j) \subset K(N^k)$ (Lemma 6.5). So $K(N^j) \subset Y'$ and we can chose an open set U' in Y' for the *i*-renormalization of N^k . Since the conjugacy σ is defined on $Y', f_c^i: \sigma(U') \to \sigma(N^j(U'))$ is a renormalization of f_c .

Proof of Point 3 of Proposition 8.3. Case 3) is exactly Theorem 4. Indeed, the orbit of the free critical point x_0 is in the union of $N^r(K_c)$ for $r \ge 0$. Since K_c is periodic, if the orbit of x_0 accumulates \overline{B}_i then $N^r(K_c) \cap \overline{B}_i \neq \emptyset$ for some $r \ge 0$ and so K_c crosses \overline{B}_i (by periodicity).

Proof of Point 4 of Proposition 8.3. Case 4) follows from Proposition 8.2 and Proposition 8.1 (as well as Theorem 4). The reasons are that ∂B_i crosses K_c at most on β_c (Corollary 6.4) and since f_c is renormalizable (Lemma 8.8) β_c is not accumulated by the orbit of x_0 . More precisely, if K' corresponds to

164

the filled Julia set of a renormalization of f_c , Theorem 7.10 of [McM1] insures that K' does not contain $\beta(f_c)$. Hence the orbit $f_c^n(K')$ (for $n \ge 0$) also avoid $\beta(f_c)$ (since $\beta(f_c)$ is fixed and K' periodic). Thus using the conjugacy σ , we see that the point β_c is not in the closure of the orbit of $N^{kn}(x_0)$, nor is it in the closure of the orbit of x_0 after Corollary 6.6.

THEOREM 1. There exist cubic Newton methods possessing a Cremer point and having a locally connected Julia set.

Proof. Note that these maps are genuine Newton methods.

The idea is to find a cubic Newton map, at least twice renormalizable, possessing a Cremer point. By Proposition 8.3 (4) the Julia set will be locally connected. For this, we use the fact that in the parameter plane of cubic Newton method, there are copies of the Mandelbrot set $M = \{c \mid J(f_c) \text{ is connected}\}$ (here $f_c(z) = z^2 + c$). Indeed, considering the family of Newton maps N_{λ} associated to $P_{\lambda}(z) = (z-1)(z-\frac{1}{2}+\lambda)(z-\frac{1}{2}-\lambda)$ for $\lambda \in \mathbf{C}$, Douady and Hubbard ([D-H2, Chap. VI]) proved that there exists a subset M_0 of **C** and a surjective map $\chi: M_0 \to M$ such that for all $\lambda \in M_0, N_\lambda$ is renormalizable at $x_0 = 0$ and the renormalization is quasi-conformally conjugated in a neighborhood of the Julia set to the quadratic polynomial $z^2 + \chi(\lambda)$. Now since Naïshul's result (see [N]) asserts that a Cremer point for $z^2 + \chi(\lambda)$ will give a Cremer point for N_{λ} it is enough to find a quadratic polynomial $z^2 + c$ which is renormalizable and possesses a Cremer point. For this we use the existence of a copy M'of M strictly contained in M (see for example Theorem 5 of [D-H2]), i.e. a surjective map $\chi': M' \to M$ which gives the straightening parameter. Then if c is a parameter for which $z^2 + c$ has a Cremer point, $(\chi')^{-1}(c)$ does possess a Cremer point and is renormalizable.

THEOREM 2. Every infinitely renormalizable cubic Newton method has a locally connected Julia set.

Proof. This is a direct consequence of Proposition 8.3(4). Indeed, the Julia set of an infinitely renormalizable quadratic polynomial has empty interior (see [McM1, Th. 8.1]). Hence the Newton method N cannot have a Siegel point.

COROLLARY 3. There exist rational maps with locally connected Julia set and wandering continuum within.

From [Le], a continuum $K \subset J$ is called *wandering* for f if for any $n \neq m$ nonnegative, $f^n(K) \cap f^m(K) = \emptyset$. Recall that for polynomials $g_c = z^l + c$ with connected Julia set G. Levin proved that J is locally connected if and only if no continuum $K \subset J$ is wandering.

P. ROESCH

Proof. There exist quadratic polynomials $f_c(z) = z^2 + c$ such that f_c is infinitely renormalizable and $J(f_c)$ is not locally connected (see [M2]). For this polynomial, nested Julia sets J_i (containing 0 and obtained by renormalizations of f_c) satisfy $J_{\infty} = \bigcap_{i=1}^{\infty} J_i \supseteq \{0\}$ (see [Le, McM1]). Since J_i has "periods" increasing with i, J_{∞} is wandering. As in the proof (above) of Theorem 1, a Newton method N such that $\chi(N) = c$ provides an example.

8.2. Proof of Case 2) of Proposition 8.3. We recall the notation 6.2: $N^k: Y' \to Y$ denotes the (only) renormalization of N, K_c the filled Julia set, σ the conjugacy to f_c and finally the fixed point $\beta(f_c)$ with preimage $\beta'(f_c)$.

After Lemma 7.9 the Fatou components are the iterated preimages of the B_i (excepted in the geometrically finite case treated by [D-H1, M2, T-Y]). So using Proposition 8.1 and Theorem 6 it remains to prove that only finitely many of these components have diameter greater than ε . If $\overline{B}_i \cap K_c = \emptyset$ this is direct from Proposition 8.2 since the orbit of x_0 is in the orbit of K_c (see the proof of Proposition 8.2.3). This is always the situation of \overline{B}_3 .

Hence we assume that $\overline{B}_1 \cap K_c \neq \emptyset$ (the proof is similar for B_2) and we show that the set of preimages of \overline{B}'_1 has diameters tending to 0. It will then also be the case with \overline{B}_1 since $N^{-1}(\overline{B}_1) = \overline{B}_1 \cup \overline{B}'_1$. Recall that in this case $\overline{B}_1 \cap K_c = \beta_c$ and $\overline{B}'_1 \cap K_c = \beta'_c$ (see Lemma 6.3).

Definition 8.9. A set \mathcal{E} of connected parts of $\widehat{\mathbf{C}}$ is said to have diameters tending to 0 if for every $\varepsilon > 0$ only a finite number of elements of \mathcal{E} have diameter greater than ε .

LEMMA 8.10. There exists an arbitrarily small neighborhood Q of $\beta'(f_c)$ such that the set of iterated preimages of Q by f_c has diameters tending to 0.

Proof. Since this neighborhood is constructed as a puzzle piece for f_c , so we have to recall some facts. On $\mathbb{C} \setminus K(f_c)$ the polynomial f_c is conjugated to the map $z \mapsto z^2$ in $\mathbb{C} \setminus \overline{\mathbb{D}}$. This conjugacy defines external rays and equipotentials (respectively) as the images of rays and circles of $\mathbb{C} \setminus \overline{\mathbb{D}}$. The construction of Yoccoz' puzzle (see also [M2] and [Hu]) requires the second fixed point $\alpha(f_c)$ also to be repelling (which is the case here by assumption). Then there exists q external rays, of angles $2^i p/(2^q - 1)$, $1 \leq i \leq q$, converging to it and forming, with an external equipotential, a graph for f_c . Assuming f_c is not geometrically finite, this graph will be admissible (see Definition 2.13 where X is the disc bounded by the equipotential).

Then to every point $z \in J(f_c)$ we associate $P_r^*(z)$ the union of the puzzle pieces of depth r whose closure contains z. These closed neighborhoods of zsatisfy, after Theorem 2 of [M2, p. 14] (or Theorem 5.7.a of [Hu, p. 483]), that

(*)
$$\bigcap_{r \ge 0} \overline{P}_r^*(z) = \{z\}.$$

Hence, since $\beta(f_c)$ is not on the graph, it belongs to a piece, say P, of arbitrarily large depth and by (*) of arbitrarily small size.

Assume (by contradiction) that there exists $\varepsilon > 0$ and a sequence P_i of the iterated preimages of P by f_c whose depth increases and diameter stands greater than ε . Up to extraction, we can find a sequence $x_i \in P_i$ of preimages of $\beta(f_c)$ converging to a point $x \in K(f_c)$. Hence for each depth r, there exists i large such that the points x_i belong to $P_r^*(x)$. Thus, $P_i \subset P_r^*(x)$ (since the pieces are disjoint or nested). Finally the preimage of P containing $\beta'(f_c)$ gives the announced neighborhood which contradicts (*).

We choose a neighborhood Q sufficiently small so that $Q \subset \sigma(Y')$ and decompose B'_1 in disjoint parts: $B'_1 = Q'_1 \amalg R'_1$, where Q'_1 is the connected component of $B'_1 \cap \sigma^{-1}(Y')$ whose closure contains β'_c (it is unique since $\partial B'_1$ is a Jordan curve) and $R'_1 = B'_1 \setminus Q'_1$.

Since \overline{R}'_1 is disjoint from $\bigcup_{0 \le i \le k} N^i(K_c)$, that contains the post-critical set of N, the set of its iterated preimages have diameters tending to 0 (Proposition 8.2).

We now concentrate on \mathcal{Q} the set of preimages of Q'_1 :

Let \mathcal{Q}^i $(0 \leq i \leq k-1)$ be the set of iterated preimages of Q'_1 by N whose closure touches $N^i(K_c)$. If U is a connected component of \mathcal{Q} not in any \mathcal{Q}^i , then for some $j \geq 0$, $N^j(U)$ belongs to a component of $N^{-1}(N^i(Y'))$ disjoint from the post-critical set; that is, U is in

$$Y^{-1} = N^{-1} \left(\bigcup_{0 \le i \le k-1} N^i(Y') \right) \setminus \bigcup_{0 \le i \le k-1} N^i(Y').$$

As above $\overline{Y^{-1}}$ is disjoint from $\bigcup_{0 \le i \le k} N^i(K_c)$ so its iterated preimages have diameters tending to 0 (Proposition 8.2), and also $\mathcal{Q} \setminus \bigcup_{0 \le i \le k-1} \mathcal{Q}^i$ have diameters tending to 0. Hence it remains to show it for $\bigcup_{0 \le i \le k-1} \mathcal{Q}^i$.

Remark 8.11. 1) If g denotes the restriction of $N^k \colon Y' \to Y$ then for Y small enough \mathcal{Q}^0 coincides with the set of iterated preimages of Q'_1 by g. This follows directly from the fact that if Y is small enough, the sets $(N^i(Y))_{0 \leq i \leq k-1}$ are disjoint (Corollary 6.6).

2) Every connected component of \mathcal{Q}^i is also in $(N^{k-i})^{-1}(\mathcal{Q}^0)$.

After Remark 8.11.1), every preimage of Q by f_c is the image by σ of a component of Q^0 . Hence, Lemma 8.10 and Lemma 8.12 (below) insure that Q^0 has diameters tending to 0. Finally, Remark 8.11.2) and Lemma 8.12 allow us to conclude that the Q^i also have diameters tending to 0.

LEMMA 8.12. Let Z, Z' be two open sets of $\widehat{\mathbf{C}}$, $\rho: Z' \to Z$ a (topological) ramified covering map of degree d and \mathcal{E} (resp. \mathcal{E}') the set of the relatively compact connected open sets of Z (resp. of Z'). If \mathcal{E} has diameters tending to 0 and if ρ sends every open set of \mathcal{E}' inside an open set of \mathcal{E} , then \mathcal{E}' also has diameters tending to 0.

Proof. The proof is by contradiction. We assume that there exists $\varepsilon > 0$ and U'_i of \mathcal{E}' with diameter greater than ε . Hence, in each U'_i , one can find d+1 points z_i^j , $1 \le j \le d+1$, at (mutual) distance greater than $\varepsilon/(d+1)$. One extracts d+1 subsequences converging to points z^j at distances at least $\varepsilon/(d+1)$. We show now that those d+1 points have the same image under ρ .

Let U_i be the sequence of images $\rho(U'_i)$ and take any n > 0. For *i* large enough, the diameter of U_i is less than 1/n. Hence, the (mutual) distances between the points $\rho(z_i^j)$ are less than 1/n. Taking the limit over *i*, one sees that the mutual distances between the images $\rho(z^j)$ are less than 1/n. Since this is the case for any n > 0, the points $\rho(z^j)$ collapse, which contradicts the fact that ρ is of degree *d*.

8.3. Proof of Case 1) of Proposition 8.3. This part is very technical and deserves only the *a priori* easiest case: the nonrenormalizable case. We use the theory of Yoccoz' puzzle and precisely Theorem 2.15, not only for ∂B_i as in Section 7 but for the whole Julia set. Therefore we need the construction of a new kind of graph. In opposition to Section 7, the difficulty here is to satisfy the hypothesis of Theorem 2.15, since for the conclusion we choose the good one, namely the nonrenormalizable case.

Definition 8.13. Let $\eta, \tau \in G \cap \mathbf{Q}, \zeta \in G \cap [\alpha, 2\alpha]$ be a nondyadic rational number and L be the articulated ray constructed in Proposition 4.3 stemming from B_2 with angle $-\zeta/4$. We define a new type of graph:

$$III(\zeta,\eta,\tau) = \partial X \cup \left(X \cap \left(\bigcup_{j \ge 0} (N^j(L) \cup \overline{R}_3(2^j/7) \bigcup_{t=\eta,\tau} (\overline{R}_1(2^jt) \cup \overline{R}_2(-2^jt))) \right) \right).$$

The type III graph corresponds to the type II graph with in addition the orbit of $\overline{R}_1(\tau)$ and $\overline{R}_2(-\tau)$. For the graph of type II, the points surrounded by a nondegenerate annulus of depth 0 were limitated by the preimages of L and of $N^3(L)$. Here while taking the preimages of $R_1(\tau), R_2(-\tau)$ we enlarge this zone almost up to $N^3(L)$.

Since the type III graphs (as type II) are dis-symmetric, we consider the symmetric one: $III^*(\zeta, \eta, \tau)$ constructed as $III(\zeta, \eta, \tau)$ but instead of the images $N^j(L)$ we take the images $N^j(L^*)$, L^* being the articulated ray (symmetric to L) defined as follows:

PROPOSITION 8.14. Let y be the landing point of $R_3(6/7)$ and $\zeta \in G$ $\cap]\alpha, 2\alpha[$ be a nondyadic angle. There exists a unique articulated ray L^* stemming from B_1 with angle $\zeta/4$ such that:

$$N^{3}(L^{*}) = L^{*} \cup \overline{R}_{1}(\zeta) \cup \overline{R}_{2}(-\zeta) \cup \overline{R}_{1}(2\zeta) \cup \overline{R}_{2}(-2\zeta)$$

and L^* converges to y.

The proof is the same as that of Proposition 4.3.

As in Section 5 we will only consider graphs of type I, III, III^* with angles of the form (\diamond):

$$\begin{cases} \theta = 1 - \frac{1}{2^p - 1}, \eta = 1 - \frac{1}{2^n - 1}, \zeta = \frac{1}{2^l} \left(1 - \frac{1}{2^m - 1} \right), \tau = \frac{1}{2^l} \left(1 - \frac{1}{2^r - 1} \right) \\ \text{such that} : \frac{1}{2} > \frac{\eta}{2} > 2^{l-1}\zeta > 2^{l-1}\tau > 2^l\tau + \frac{1}{2} > 2^{l-1}\alpha > \frac{1}{4} \text{ and } \theta - 1/2 > \alpha. \end{cases}$$

For this it is enough to take p large, l such that $1/2 > 2^{l-1}\alpha \ge 1/4$ and n > m > r, with r large enough so that $\frac{1}{2} > \frac{1}{2}(1-\frac{1}{2^r-1}) > 2^{l-1}\alpha$. In particular we obtain that $\tau, \zeta \in]\alpha, 2\alpha[$.

PROPOSITION 8.15. There exist an integer δ and a finite number of puzzles of type I, III and III^{*} such that x_0 , resp. every point of the Julia set, is surrounded by each, resp. one, of these puzzles and at depth less than δ .

Proof. We will refer intensively to the proof of Proposition 5.4, in particular to Remark 5.2 and 5.3 and to the notation namely of the articulated rays $L^0 = L$, $L^1 \subset N(L)$ and $L^2 \subset N^2(L)$.

Step1. For ζ, η, τ satisfying (\diamond) and for r sufficiently large, the graphs $III(\zeta, \eta, \tau)$ and $III^*(\zeta, \eta, \tau)$ are puzzles and surround x_0 .

Such graphs are admissible since the critical orbit cannot have arbitrarily large period ($\geq r$). Then the argument goes as in Step 5 of Proposition 5.4. After Lemma 3.14, x_0 belongs to the bounded connected component of $\widehat{\mathbf{C}} \setminus \gamma(\frac{2^l \tau + 1/2}{2^{l-1}}, \frac{\eta}{2^{l+1}})$ since $\alpha < \frac{2^l \tau + 1/2}{2^{l-1}} < \eta/2^l < 2\alpha$ by (\diamondsuit).

Notation. Let L', resp. L'', L'^* and L''^* be the articulated rays stemming from B_1 , resp. B'_2 , B_2 and B'_1 such that $N(L') = L^2$, resp. $N(L'') = L^1$, $N(L'^*) = L^{2^*}$ and $N(L''^*) = L^{1^*}$.

Step 2. The bounded component of

$$X' \setminus \left(L' \cup L'' \cup \overline{R}_1(\tau/2) \cup \overline{R}'_2(-\tau) \cup \overline{R}_3(9/14) \cup \overline{R}_3(11/14) \right),$$

is a piece of depth 1 of the puzzle $III(\zeta, \eta, \tau)$, with angles satisfying (\diamondsuit) and its boundary is disjoint from $III(\zeta, \eta, \tau)$ (similarly with $III^*(\zeta, \eta, \tau), L'^*, \ldots$).

The image of this connected component is the piece of depth 0 located between N(L), $N^2(L)$, $R_1(\tau)$, $R_2(-\tau)$ etc. Its boundary consists at depth 0



Figure 14: On the left useful parts of the graph $III(\zeta, \eta, \tau)$ and some dashed preimages. On the right U and in dashed V.

and 1 of rays of angles in B_1 , resp. B'_2 , strictly between 0 and $\zeta/2$, resp. between $-\zeta$ and $-\zeta/4$. Moreover, at depth greater than 1, L'' and L^2 cannot cross, else N(L) and L would cross at depth greater than 0.

Notation. For $(\theta_j, \zeta_j, \eta_j, \tau_j)_{j=1,2}$, let $\Omega(\theta_1)$ be the bounded component of $\widehat{\mathbf{C}} \setminus \bigcup_{i=1,2} \overline{R}_i((-1)^{i+1}\theta_1 + 1/2) \cup \overline{R}_i((-1)^{i+1}\theta_2/4) \cup \overline{R}'_i((-1)^{i+1}\theta_2/2) \cup \overline{R}'_i(0)$ and for j = 1, 2, let U_j be the unbounded connected component of the complement of $L''_j \cup L'^{*''}_j \cup \overline{R}_3\left(\frac{\pm 5}{14}\right) \bigcup_{i=1,2} \overline{R}_i\left(\frac{(-1)^{i+1}\eta_j+1}{2}\right) \cup \overline{R}_i\left(\frac{(-1)^{i+1}\tau_j}{2}\right) \cup \overline{R}'_i\left((-1)^{i+1}\tau_j\right).$ $(L_j$ is the articulated ray constructed with ζ_j etc....)

Step 3. For $\theta_1 > \theta_2$ and $\eta > \zeta_1 > \tau_1 > \zeta_2 > \tau_2$ satisfying (\diamondsuit) with the same l, the points of $J(N) \cap (U_1 \cup U_2 \cup \Omega(\theta_1))$ are surrounded by a nondegenerated annulus of depth less than 2 for $III(\zeta_j, \eta, \tau_j)$ or $III^*(\zeta_j, \eta, \tau_j)$ or $I(\theta_j)$ with $j \in \{1, 2\}$.

The unbounded pieces of depth 0 and 1 are the same for the graphs of type II or III. By Step 4 of Proposition 5.4 the unbounded piece of depth 1 is the central disc of a nondegenerated annulus of depth 0 (defined by $III(\zeta_j, \eta, \tau_j)$) with $j \in \{1, 2\}$). Using Step 2 and the similar result for $III^*(\zeta_j, \eta, \tau_j)$ we obtain the part $U_1 \cup U_2$. Indeed, the points which are on $J \cap U_1 \cap N^{-1}(III(\zeta_1, \eta, \tau_1))$ are in $U_2 \setminus N^{-1}(III(\zeta_2, \eta, \tau_2) \cup III^*(\zeta_2, \eta, \tau_2))$. Moreover the sets $X' \setminus \gamma(\theta_j + \eta)$

 $1/2, \theta_j/4$) and $X' \setminus \bigcup_{i=1,2} \overline{R}'_i((-1)^{i+1}\theta_j/2) \cup \overline{R}'_i(0)$ are unions of depth 1 pieces for $I(\theta_j)$, compactly contained in a depth 0 piece (proof of Proposition 5.4, Step 2). Hence by taking two values of the angles $\theta_1 > \theta_2$ we include in the domain the points of the graph and obtain $\Omega(\theta_1)$.

Step 4. The points of $J(N) \setminus (U_1 \cup U_2 \cup N^{-1}(U_1 \cup U_2))$ are in the bounded connected component of $\widehat{\mathbf{C}} \setminus \widetilde{C}$ where $\widetilde{C} = \gamma(\frac{\tau_1}{2}, \frac{1+\eta}{2})$.

Let V_j be the bounded connected component of $N^{-1}(U_j)$ intersecting ∂B_3 . We will prove the statement with V_j instead of $N^{-1}(U_j)$. For this we study the intersections of the boundary of U_1, V_1, U_2, V_2 , concentrating on the rays and articulated rays by depth. As the picture is symmetric, it is enough to study the boundaries on one side for instance near $B_2, B'_1 \dots$. At depth 0, the rays are exactly those involved in \widetilde{C} . At depth greater than 3 the boundaries do not cross, else by iteration the articulated rays L^{*0} and L^2 would also cross at depth greater than 0. Hence V_j covers ∂U_j at depth greater than 3 since the articulated ray of the boundary of V_j , resp. U_j , converges to the same points as $R_3(9/28)$, resp. $R_3(5/14)$.

At depth 1, U_2 , resp. V_1 , contains at least in B'_1 the part between the angles $\zeta_2/2$ and τ_2 , resp. 0 and $\tau_1/2$ so that V_1 contains $R'_1(\zeta_2/2)$ since $\tau_1 > \zeta_2$. At depth 2 the situation is similar for the rays touching the previous one. Moreover, U_1 , resp. V_2 , contains at least the part between the angles $-\zeta_1/2$ and 1, resp. $-\tau$ and $-\zeta_2/2$ so that V_2 covers U_1 at least on the ray of angle $-\zeta_1/2$ since $\zeta_1 > \zeta_2$. At depth 3 the situation is identical for the consecutive rays. Finally, the next rays of depth 3 have angle $\zeta_j/2$ in ∂U_j and ζ_j in ∂V_j so that U_j covers V_j along this ray.

Step 5. There exist θ_1 and $\delta \in \mathbf{N}$ such that $J(N) \subset N^{-\delta}(\Omega(\theta_1) \cup U_1 \cup U_2)$.

The inverse image of $U_1 \cup U_2 \cup V_1 \cup V_2$ containing $R_1(1/2)$ covers the bounded complement of $\bigcup_{i=1,2} \overline{R}_i(1/2) \cup \overline{R}_i((1+(-1)^{i+1}\eta)/4)$. Choosing θ_1 such that $1/2 > \theta_1 - 1/2 > (\eta+1)/4$, we take p large, $N^{-1}(U_1 \cup U_2 \cup V_1 \cup V_2) \cup \Omega(\theta_1)$ covers Q_1 the bounded connected component of $\widehat{\mathbf{C}} \setminus \bigcup_{i=1,2} \overline{R}_i(1/2) \cup \overline{R}_i(\theta_1/4)$.

If $\widehat{\mathbf{C}} \setminus \gamma(1/2, \alpha/2)$, Q_2 denotes by the bounded connected component, then every point of Q_2 is sent by an iterate of N (less than l + 1) into Q_1 (to see this, it is enough to cut the component in "sectors" of angles $(2^t \alpha/2, 2^{t+1} \alpha/2)$). Moreover, $\Omega(\theta_1)$ contains the bounded complement of $\bigcup_{i=1,2} \overline{R}'_i(0) \cup \overline{R}'_i(1/2)$ as well as the rays $\overline{R}'_i(1/2)$ (because $\theta_2/2 < 1/2$). Hence $Q_2 \cup \Omega(\theta_1) \supset Q_3$ the bounded complement of

$$\cup_{i=1,2}\overline{R}_i((-1)^{i+1}\alpha/2)\cup\overline{R}_i(1/2)\cup\overline{R}'_i((-1)^{i+1}\alpha)\cup\overline{R}'_i(0).$$

By Step 4 and since $\tau_1 > \alpha$, the union $Q_4 = \Omega(\theta_1) \cup \bigcup_{i=1,2} U_i \cup N^{-1}(U_i) \cup Q_i$ contains the unbounded complement of $\bigcup_{i=1,2} \overline{R}_i((1+\eta)/2) \cup \overline{R}_i(1/2)$ (excepted maybe some points outside J).

Let q be the first integer such that $1 - 1/2^q > (\eta + 1)/2$. Every bounded t < q sent by N^t into Q_4 .

Université de Lille 1, Villeneuve d'Ascq, France Current address: Université Toulouse 1 Sciencies Sociales, Toulouse, France E-mail address: roesch@univ-tlse1.fr

References

- [A] L. V. ALHFORS, Lectures on Quasi-Conformal Mappings, Wadsworth & Brook/Cole, Advanced Books & Software, Monterey, 1987.
- [B] P. BLANCHARD, Complex analytic dynamics on the Riemann sphere, Bull. Amer. Math. Soc. 11 (1984), 85–141.
- [B-L]A. BLOKH and G. LEVIN, An inequality for laminations, Julia sets and growing trees, Ergodic Theory Dynam. Systems 22 (2002), 63–97.
- [B-H] B. BRANNER and J. H. HUBBARD, The iteration of cubic polynomials, Acta Math. **169** (1992), 229–325.
- [CG]L. CARLESON and T. GAMELIN, Complex Dynamics, Springer-Verlag, New York, 1993.
- [C]H. CREMER, Zum Zentrumproblem, Math. Ann. 98 (1927), 151-163.
- A. DOUADY and J. H. HUBBARD, Etude dynamique des polynômes complexes, Pub-[D-H1] lications mathématiques d'Orsay, 1984.
- , On the dynamics of polynomial-like mappings, Ann. Sci. École Norm. Sup. [D-H2] **18** (1985), 287–343.
- [D]A.DOUADY, Disques de Siegel et anneaux de Herman, Sém. Bourbaki 39 (1986-1987), 151-172.
- [F]D. FAUGHT, Local connectivity in a family of cubic polynomials, Ph.D. Thesis, Cornell University, 1992.
- [G-M] L. GOLDBERG and J. MILNOR, Fixed points of polynomial maps. II. Fixed point portraits, Ann. Sci. École Norm. Sup. 26 (1993), 51-98.
- [G-Sm] J. GRACZYCK and S. SMIRNOV, Non-uniform hyperbolicity in complex dynamics I, II, preprint, 1999.
- [G-Sw] J. GRACZYK and G. SWIATEK, The Real Fatou Conjecture, Ann. of Math. Studies 144, Princeton Univ. Press, Princeton, NJ (1998).
- [He] J. HEAD, The combinatorics of Newton's method for cubic polynomials, Ph.D. Thesis, Cornell University, 1988.
- J. H. HUBBARD, Local connectivity of Julia sets and bifurcation loci: three the-[Hu] orems of J.-C. Yoccoz, in Topological Methods in Modern Mathematics, 467–511 (L. R. Goldberg and A. V. Phillips, eds.), Publish or Perish Houston, TX, 1993.
- [J]Y. JIANG, Infinitely renormalizable quadratic polynomials, Trans. Amer. Math. Soc. **352** (2000), 5077–5091.
- [K]J. KIWI, Real laminations and the topological dynamics of complex polynomials, Adv. Math. 184 (2004), 207-267.
- [Le] G. LEVIN, On backward stability of holomorphic dynamical systems, Fund. Math. **158** (1998), 97–107.

- [L-vS] G. LEVIN and S. VAN STRIEN, Local connectivity of the Julia set of real polynomials, Ann. of Math. 147 (1998), 471–541.
- [Ly] M. LYUBICH, Dynamics of quadratic polynomials, I–II, Acta Math. 178 (1997), 185– 297.
- [L-M] M. LYUBICH and Y. MINSKY, Laminations in holomorphic dynamics, J. Differential Geom. 47 (1997), 17–94.
- [McM1] C. MCMULLEN, Complex Dynamics and Renormalization, Ann. of Math. Studies 135, Princeton Univ. Press, Princeton, NJ, 1994.
- [McM2] _____, Renormalization and 3-Manifolds which Fiber over the Circle, Ann. of Math. Studies 142, Princeton Univ. Press, Princeton, NJ, 1996.
- [M1] J. MILNOR, Dynamics in One Complex Variable, Vieweg, 1999, 2nd edition, 2000.
- [M2] _____, Local Connectivity of Julia Sets: Expository Lectures, in The Mandelbrot Set, Theme and Variations (Tan Lei, ed.), London Math. Soc. Lecture Note Series 274, 67–116, Cambridge Univ. Press, Cambridge, 2000.
- [N] V. A. NAĬSHUL', Topological invariants of analytic and area-preserving mappings and their application to analytic differential equations in C^2 and CP^2 , Trans. Moscow Math. Soc. 42 (1983), 239–250.
- [Pe1] C. L. PETERSEN, Local connectivity of some Julia sets containing a circle with an irrational rotation, Acta Math. 177 (1996), 163–224.
- [Pe2] _____, On the Pommerenke-Levin-Yoccoz inequality, Ergodic Theory Dynam. Systems 13 (1993), 785–806.
- [PM1] R. PÉREZ-MARCO, Solution complète au problème de Siegel de linéarisation d'une application holomorphe au voisinage d'un point fixe (d'après J.-C. Yoccoz), Sém. Bourbaki 753 (1991–1992).
- [PM2] _____, Sur les dynamiques holomorphes nonlinearisables et conjectures de V. I. Arnold, Ann. Sci. Norm. Sup. 26 (1993), 565–644.
- [Po] C. POMMERENKE, Boundary Behaviour of Conformal Maps, Springer-Verlag, New York, 1992.
- [Ra] M. RABII, Local connectedness of the Julia set of the family $z^m(z^n b)$, Ergodic Theory Dynam. Systems 18 (1998), 457–470.
- [Ri] J. RIVERA-LETELIER, Weak hyperbolicity on periodic orbits for polynomials, C. R. Math. Acad. Sci. Paris 334 (2002), 1113–1118.
- [Ro] P. ROESCH, Puzzles de Yoccoz pour les applications à allure rationnelle, L'Enseign. Math. 45 (1999), 133–168.
- [Sh] M. SHISHIKURA, The connectivity of the Julia set of rationnal maps and fixed points, preprint IHES, Bures-sur-Yvette, 1990.
- [So1] D. SORENSEN, Describing quadratic Cremer point polynomials by parabolic perturbations, *Ergodic Theory Dynam. Systems* 18 (1998), 739–758.
- [So2] _____, Infinitely renormalizable quadratic polynomials, with non-locally connected Julia set, J. Geom. Anal. **10** (2000), 169–206.
- [Su] D. SULLIVAN, Quasiconformal homeomorphisms and dynamics I. Solution of the Fatou-Julia problem on wandering domains, Ann. of Math. 122 (1985), 401–418.
- [T1] L. TAN, Branched coverings and cubic Newton maps, Fund. Math. 154 (1997), 207–260.
- [T2] _____, Sur quelques aspects de l'itération des fractions rationnelles, Habilitation à diriger des recherches, Université Claude Bernard Lyon I, 1997.

P. ROESCH

- [T-Y] L. TAN and Y. YIN, Local connectivity of the Julia set for geometrically finite rational maps, *Science in China* (Serie A) **39** (1996), 39–47.
- [W] G. WHYBURN, Analytic Topology, AMS Colloq. Publ. 28, 1942.
- [Y] J.-C. YOCCOZ, Théorème de Siegel, nombres de Bruno et polynômes quadratiques.
 Petits diviseurs en dimension 1, Astérisque No. 231 (1995), 3–88.

(Received July 12, 2004) (Revised October 11, 2005)

174